Does Innovation Promote Performance? Evidence from Developed Economies During Crises

Ameni Hamdouni^{1*} Dorra Dallali² Marwa Marzouki² Nadia Ben Yedder³

1.University of Tunis-Higher Institute of Management of Tunis, n° 41 Liberty Avenue, Bouchoucha, Bardo 2000, Tunisia

2. University of Tunis-El Manar-Faculty of Economics Sciences and Management of Tunis, B.P. n° 94 -

Rommana, Tunis 1068, Tunisia

3. University of Manouba- Higher Business School of Tunis ESCT, Manouba university campus - 2010 Tunisia * E-mail of the corresponding author: Amenihamdouni.isg@gmail.com

Abstract

This study examines the relationship between firm performance and innovation levels of Information Technology and Communication (ICT), covering 886 firms from five developed countries during the period 2000-2020. The results indicate a positive association between the accessibility and/or the use of ICTs and firms' innovation performance. More precisely, the positive relationship between ICTs and firm performance is highly pronounced in small-sized firms and strongly related to R&D decisions in large firms. However, during the recession period, the relationship between performance and innovation level is affected differently by the financial crisis, leading to contradictory results. Interestingly, the pandemic crisis has had a positive impact on the firm's performance-innovation relationship. Despite high rates of ICT diffusion, the results of performance and innovation cannot be considered universal due to the significant differences between countries. Our findings may contribute to the literature by highlighting how variations in ICT impact firms' innovation performance across countries, particularly during crises.

Keywords: Information and Communication Technology, Performance, Innovation, Financial Crisis, Pandemic Crisis.

DOI: 10.7176/RJFA/14-19-03 **Publication date:** November 30th 2023

1. Introduction:

In recent times, there has been a widespread belief among scientists, researchers, and managers that firms must embrace digital innovations. These innovations are defined as "the change or creation of products and innovation processes that result from the new combinations of digital and physical components enabled by digital technology" (Liu et al., 2022). It represents; "the driving force for development"; (Schumpeter, 1934) and is increasingly employed as the main concept for governing countries (Wang and Qi, 2021). The significance of information and communication technologies (ICTs) in creating new development avenues, enhancing new driving forces of development, and improving innovation performance has become increasingly the main challenge for firms (Ani and Ofikwu, 2019; Ren-Jye, Hsien, and Keisuke, 2018; Zhou, Gao, and Chimhowu, 2019). Indeed, there has recently been a focus on the rapid development of technology and the increasing use of ICTs. ICTs have fundamentally altered the global landscape, changing economic structures and leading to the development of new types of organizations and social networks (Lechman and Marszk, 2019).

The use of ICTs is critical in achieving innovation since they contribute to all aspects of the innovation process. Numerous studies suggest that ICTs can enhance innovation performance (Shen and Yuan, 2020; Wang and Qi, 2021; Zhu et al., 2021). In the business environment, technology is seen as a vital necessity (Aragón-Sánchez and Sánchez-Marín, 2005; Rivard, 2000), and it is increasingly becoming an integral part of a firm's strategy. The use of ICTs is a key strategy that stimulates innovation and firm performance (Cuevas-Vargas and Parga-Montoya, 2022) and is the primary driving force for development. Therefore, it is essential to consider the impact of a firm's use of technology, particularly when aimed at strengthening innovation activities. ICTs play a crucial role in enabling companies to promote and innovate. According to Guillemette and Paré (2012), the literature is increasingly focused on determining the impact of ICT on firm performance. Similarly, Rousseau et al., (2016) contend that innovation helps firms to enhance performance levels and is a major factor contributing to a firm's growth and economic development, explaining why some firms outperform others. Yeo and Grant (2019) suggest that besides to improving performance, firms can increase sales, market share, and profits through innovation, which can enhance positively their overall performance (De Guimarães et al., 2016; Centobelli, Cerchione, and Singh 2019). Similarly, Vu, Hanafizadeh, and Bohlin (2020) state that causality runs from ICT to economic growth leading to development at the economic as well as the microeconomic levels (Fernández-Portillo, Almodóvar-González, and Hernández-Mogollón, 2020). This is likely to improve firm processes, products, sales, and ultimately increase profits. This predicts that ICTs will play an incremental role in improving the quality of economic development and innovation performance (Ani and Ofikwu, 2019).

From a crisis perspective, the innovation literature highlights that external competitive pressures significantly affect firms' innovative efforts (Klewitz and Hansen, 2014). Currently, two major global crises - the 2007 financial crisis and the COVID-19 pandemic- are in the spotlight for managers and policymakers. The financial crisis led to an extreme environment characterized by high levels of instability, volatility, uncertainty, and ambiguity. This resulted in limited business opportunities, and forced firms to reduce their innovation resources and efforts (Archibugi, Filippetti, and Frenz, 2013; Colombo et al., 2016). On the other hand, The COVID-19 pandemic has caused considerable disturbance across the global economic landscape, creating an environment of uncertainty that is disruptive and challenging for organizations (Biron et al., 2021).

Innovation is a crucial strategy for firms and societies to survive and recover from the pandemic crisis (Wenzel, Stanske, and Lieberman, 2020). Introducing new products, processes, and services that can overcome the challenges posed by the crisis will be crucial to firms' survival and recovery (Roper and Turner, 2020). Additionally, innovation can help firms to maintain their competitiveness during and after the pandemic crisis (Lee and Trimi, 2021; Wang et al., 2020). Therefore, despite the evidence exposed by these studies, there is a lack of consensus about the relationship between crises and firms' performance and innovation. To address this issue, our study investigates the consequences of the financial and pandemic crises on firms' performance and innovative ability in five developed countries from 2000 to 2020. We perform several robustness checks using a cross-country analysis to enhance the credibility of our findings.

Our study makes important contributions to the innovation management literature by examining the relationship between innovation and firm performance during more than one crisis (financial and pandemic crises). Additionally, it adds important evidence that comes in response to the limited studies considering the firm's innovation-performance relationship across all sectors (except the financial sector). Furthermore, only a few previous studies considered the firm's sales proxy measure as a performance measure. Finally, we incorporate three innovation indicators including fixed broadband networks, internet users rates, and the Global Innovation Index (GII).

The present study is structured as follows: section 2 presents the theoretical background and literature review; section 3 describes the methodology, data, and summary statistics; section 4 discusses the main empirical results; section 5 presents robustness checks and additional analyses; and finally, section 6 concludes, highlights relevant limitations, and provides suggestions for future studies.

2. Background Literature: Innovation and Firm Performance:

Innovative resources play a crucial role in driving business growth by creating value and improving inclusive business performance through fostering a culture of innovation and increasing business activities (Chesbrough, 2006; Enkel and Gassmann, 2010). However, the effective and proper use of innovation can be challenging for organizations due to the ever-changing and dynamic nature of the market (Chesbrough, 2006). The availability of skilled experts, diverse assets, and external suppliers also shape the innovative landscape and present further opportunities for organizations to innovate. According to Chesbrough (2003), adopting innovation programs is a key factor for firms' success and is influenced by various factors. Firms can boost sales, market share, and profits through innovation, which promotes their overall performance (De Guimarães et al., 2016; Centobelli, Cerchione, and Singh, 2019). Recent research in the field of operational management emphasizes the importance of innovation in enhancing a firm's performance (Jackson et al., 2016; Lee, Lee, and Garrett, 2019; Lichtenthaler, 2016; Prajogo, 2016).

Innovation is widely acknowledged as a catalyst for knowledge-based economic growth and long-term economic development (Arrow, 1962; Romer, 1990). Nevertheless, innovators may fail to achieve their potential economic benefits due to the uncertainty associated with the innovation process and its market outcomes (Hottenrott and Peters, 2012). To address this issue, scientists attempt to solve these dilemmas theoretically, such as the "Profit from Innovation" framework, which elucidates why innovative companies often struggle to generate substantial economic returns from their innovation ventures (Teece, 1986).

The framework utilized in this analysis of innovation economics encompasses a broader range of factors than previous approaches (Teece, 2018, 2010). Consequently, innovation is recognized as a mechanism that facilitates firms' adaptability and flexibility in response to changing economic conditions (Dougherty and Hardy, 1996) and is closely related to firm performance (Ferreira, Coelho, and Moutinho, 2020; Karlsson and Tavassoli, 2016). Similarly, Siriram (2022) contends that firms operating in a volatile environment may face underresourcing and limited access to external markets (Zhu et al., 2019). On the one hand, firms with limited resources are often driven to innovate (Bodlaj, Kadic-Maglajlic, and Vida, 2020), and innovation can enable firms to maintain a competitive advantage and a leading role in shaping the future (Lawson and Samson, 2001). Consequently, companies must leverage both internal and external resources and actively engage with connected communities as a prerequisite for innovation, which ultimately leads to improved performance and a competitive edge (Sarasvathy, 2001). To achieve this, businesses must effectively gather and leverage valuable information from their internal and external environments to enhance their performance. In response, the ability to extend

internal knowledge from the external environment is important to complement existing skills (Medase and Abdul-Basit, 2020). Moreover, the capacity to absorb knowledge through networks is a crucial source of innovation that can enhance firm performance and ultimately promote the success of the company (Lawson and Samson, 2001). By this means, firms' innovation abilities need to expand beyond products and support a wider view of innovation (Azar and Ciabuschi, 2017; Bodlaj, Kadic-Maglajlic, and Vida, 2020; Piening and Salge, 2015). However, one of the primary challenges facing innovation research is the need to consider its constituent elements. Information and communication technologies (ICTs) play a critical role in enabling firms to innovate and remain competitive, regardless of the business environment. Thus, technology has become an indispensable requirement for innovation (Aragón-Sánchez and Sánchez-Marín, 2005; Rivard, 2000).

3. Hypothesis Development

3.1. ICT and Firm Performance:

In recent times, the world has witnessed a remarkable technological progress that has led to a shift away from conventional methods of managing firms. Instead, greater emphasis is placed on leveraging information and communication technology (ICT) to reduce operational costs, enhance efficiency, improve communication, and boost firms' performance (Gërguri-Rashiti et al., 2017). Effective management of the technological environment has become a crucial component of a firm's overall strategy (Devaraj and Kohli, 2000). An important question to consider is how information and communication technology (ICT) impacts firms' performance (Lee et al., 2016; Huang et al., 2006; Kossaï and Piget, 2014). For example, Yeo and Grant (2019) suggest that ICT can promote firms' performance. However, Schryen (2013) argues that the impact of ICT on firms' performance is not sufficient and can be better explained by considering the technological infrastructure (Torero and Von Braun, 2006). Neirotti and Pesce (2018) highlight the correlation between ICT expenditure and competitive dynamics in the Italian industry, indicating that firms tend to increase their ICT investment in high-growth industries. This is supported by Roberts et al., (2012) who suggest that munificent industries are likely to have higher ICT expenditure.

On the other hand, the effectiveness of information and communication technology depends on several factors, such as countries' infrastructure (Bankole, Osei-Bryson, and Brown, 2015; Bollou, 2006; Tan, Ng, and Jiang, 2018), manufacturing infrastructure (Archibugi and Coco, 2004) that justifies the investment decision in manufacturing (Vranakis and Chatzoglou, 2012), and/or service technologies to ameliorate significantly the firms' performance (Lee et al., 2016). In this, a study in Uganda argues that ICT implantation improves investment efficiency by 81%, reduces transaction costs by 47%, and enhances performance by 38% (Ssewanyana and Busler, 2007). In addition, it depends on ICT use motives (Zhang et al., 2018). The ICT raises firms' productivity and market share (Cardona, Kretschmer, and Strobel, 2013; Tran et al., 2014), helps firms introduce new products and services, become more customer-oriented, respond better to market changes, and develop their ability to innovate (Hall, Lotti, and Mairesse, 2013; Tran et al., 2014; Gërguri-Rashiti et al., 2017). Thus, Huang et al., (2021) find that ICTs promote firm performance, including financial profitability, marketing performance, and innovation-performance. Regarding the above development, we predict that:

H.1. The access to and the use of ICT are positively related to the firm's performance. In other words, the global innovation index is positively related to a firm's performance.

3.2. ICT and Innovation-Performance

The existing literature provides evidence that technology is a crucial driver of innovation and can facilitate the development of new and improved products and services. Effective and efficient use of ICT can promote innovation-performance (Brynjolfsson and Saunders, 2010; Bardhan, Krishnan, and Lin, 2013; Arvanitis, Loukis, and Diamantopoulou, 2013; Shen and Yuan, 2020; Wang and Qi, 2021; Zhu et al., 2021).

Gago and Rubalcaba (2007) argue that technology creates favorable conditions for the development of better products and services. Besides, Ryu and Lee (2018) use service innovation in their research and show that technology plays three roles: an innovation trigger, an innovation enabler, and a true innovator. In the same vein, Nambisan (2013) describes technology as a trigger for innovation that acts on other resources to influence innovation. Carlo et al., (2014) assume that technology plays a reinforcing role rather than a triggering or enabling role.

Shen and Yuan (2020) conducted an extensive panel data analysis of the relationship between ICT and innovation in the China Industry Business. They found a significant positive correlation between ICT use and innovation performance. The authors investigated this relationship across firms of different sizes and efficiency levels. Similarly, Wang and Qi (2021) surveyed 1,550 Chinese firms using World Bank data and found that the use of ICT led to improved innovation performance. In line with these findings, Barbieri et al., (2020) compiled a unique dataset of 480 specialized clusters in Dongguan township, Guangdong Province, and discovered that ICT-specialized firms exhibited better innovation performance. Finally, Zhu et al., (2021) used a sample of 2,700 private and 148 state-owned firms from the World Bank's 2012 survey and found that ICT had a significant

positive effect on both product and process innovation They also noted that the impact of ICT was lower than that of R&D investments.Haut du formulaire Accordingly, we provide the following hypothesis:

H.2. The access to and the use of ICT has a positive effect on innovation performance. In other words, the global innovation index (GII) has a positive impact on innovation performance.

3.3. Financial Crisis and Performance

When assessing the impact of the 2008 financial crisis on innovation, researchers have offered mixed or even conflicting evidence regarding corporate investment in R&D. For instance, Paunov (2012) discovered that many companies across eight Latin American countries paused ongoing innovation projects between 2008 and 2009 due to the crisis. Similarly, Archibugi, Filippetti, and Frenz (2013) noted a decrease in European firms' inclination to invest in innovation directly caused by the financial crisis. Nonetheless, some firms deviated from this trend and sustained their R&D investments, and regional variations were observed between northern and southern European countries. Additional studies reported that select European firms either maintained or even increased their investment in innovation after the 2008 shock (Archibugi, 2017). Notably, innovative firms faced more difficulties during the financial crisis than non-innovative firms, mainly due to limited access to external funding.

A study analyzing a sample of 16,000 firms in Germany discovered that innovative firms experienced a similar decline in sales growth as non-innovative firms after 2008. However, innovative firms experienced a more substantial decrease in investment growth than their non-innovative counterparts (Giebel and Kraft, 2019). A comparable situation occurred in the UK, where both innovative and non-innovative firms encountered difficulties in securing financing during the financial crisis. Nonetheless, innovative firms were less successful than others in raising capital and were more likely to face absolute credit rationing (Lee, Sameen, and Cowling, 2015). Similarly, a study conducted by Chung (2017) involving a sample of Korean companies demonstrated a decline in R&D investment following the 2008 crisis, further indicating that external financing for R&D investments became increasingly challenging to obtain for such firms, necessitating the use of internal financial resources. Hence, the next hypothesis was developed:

H.3. The financial crisis is negatively related to innovative performance.

3.4. COVID-19 Crisis and Performance

The global lockdown due to the corona-virus disease started in 2019. The COVID-19 pandemic has resulted in an abrupt shutdown of core businesses, a worldwide recession, and a huge crisis for firms (Galanakis et al., 2021). The sudden onset and subsequent accelerated spread of the COVID-19 pandemic have had a devastating impact on numerous businesses. In this, more than 130 firms in the United States have declared bankruptcy since March 2020, citing at least in part the COVID-19 epidemic as the principal factor causing their disappearance. In the wake of significant changes to daily lifestyles and work routines, innovation has become a vital strategy for organizations to survive and recover from the crisis (Zhong et al., 2022). The COVID-19 pandemic has significantly impacted numerous industries, and the inherent uncertainty of the markets has created ongoing challenges and barriers that companies must overcome. However, the crisis is also driving waves of innovation activity (Amankwah-Amoah, 2021). Innovation is crucial for organizations to adapt to rapid changes and remain competitive in the turbulent market environment, especially during the digital age and following the COVID-19 pandemic (Breier et al., 2021; Ebersberger and Kuckertz, 2021; Wang et al., 2020; Wenzel, Stanske, and Lieberman, 2020). As such, improving innovation performance is a key strategic response for organizations operating in a dynamic environment full of challenges and threats, such as the COVID-19 pandemic (Kotabe, Jiang, and Murray, 2017; Zhang, O'Kane, and Chen, 2020). Shin and Kang (2020) investigated the influence of expected interaction and cleanliness on perceived health risks, as well as the intention to reduce customer interaction with employees and enhance cleanliness as a crucial risk-reduction strategy, focusing on technological innovation. In the supply chain area, the COVID-19 pandemic has exposed businesses and companies to gaps in normal trends and their lasting impacts on supply chains. In this, Galanakis et al., (2021) explore potential innovations in the food sector during the COVID-19 crisis, which can be seen as the innovations with the greatest potential in the food supply chain of the new era. Thus, Nandi et al., (2021) provided information on how to make supply chains more resilient, transparent, and sustainable, added to the need to develop location, agility, and digitization features. Lee and Trimi (2021) argue that organizations need to develop a new sustainable core competency, which they call "convergence innovation". This approach is based on the exponential fusion of various objects, technologies, ideas, and strategies, and it can help organizations to stay competitive and to adapt to changing market conditions. In this regard, we provide the following hypothesis: H.4. COVID-19 crisis has positive effect on both the firms' performance and innovative abilities.

3.5. The Firm Size Moderating Role

Firms of various sizes and data management approaches may experience varying levels of government benefits

and demonstrate varying levels of performance. Typically, large firms are considered more visible and socially responsible. In contrast, small firms need to exert less pressure or gain less environmental recognition due to their lower profile. Meznar and Nigh (1995) find that larger firms exhibit lower responsiveness to society and are more resistant to external influences, which is a paradoxical finding. Meanwhile, Symeonidis (1996) notes that there is insufficient evidence to support Schumpeter's (1934) hypothesis that market dominance and large firms stimulate innovation. When certain conditions are met, such as high sunk costs for each task, economies of scale, and the ability to generate innovation rents, a positive relationship between intensity, scale, and innovation activities can emerge. However, academics believe that an organization's ability to invest in R&D depends on the firm's size and level of innovation (Baumann and Kritikos, 2016; Czarnitzki and Hottenrott, 2011; Hall, Lotti, and Mairesse, 2009).

Studies examining the link between firm size and performance yielded mixed results (Dass, 2000; Ruzzier and Ruzzier, 2015), but generally suggest that large firms benefit from economies of scale and are better positioned to engage in global venture capital (Dass, 2000; Ruzzier and Ruzzier, 2015). Bloodgood, Sapienza, and Almeida (1996) find that large firms are more likely to become global market players. So, the relationship between firm size and innovation remains inconclusive (Liu, 2014). Patacconi and Belenzon (2014) suggest that small firms tend to perform better in applied research (patents), while large firms have an advantage in baseline performance research. In addition, Lara et al., (2015) found no significant relationship between the number of patented inventions and SMEs' sales performance. Previous research indicated a positive correlation between firm size and innovation level (Jimenez and Sanz-Valle, 2011). To investigate the impact of firm size on innovation levels in the service sector, Arias-Aranda, Minguela-Rata, and Rodríguez-Duarte (2001) conducted a study on Spanish engineering consulting firms and found that firm size was strongly linked to innovation levels. Furthermore, Audretsch and Acs (1991) observed a significant positive correlation between firm size and innovation levels.

There is an ongoing debate about whether large- or small-sized firms are more able to capture the potential performance gains of innovation (Damanpour, 2010; Leiblein and Madsen, 2009). In this, Rosenbusch, Brinckmann, and Bausch (2011) advance that innovation is positively associated with the performance in a sample of small- and medium-sized firms, but did not include a comparison with large firms although small firms are, in some cases, seen as more innovative. In this view, we note that high sized firms are generally more able to translate these innovations into performance (Graebner, Eisenhardt, and Roundy, 2010). Hence, the subsequent hypothesis was advanced:

H.5. The access to and use of ICT are positively associated with performance, especially for large-sized firms. In other words, the relationship between innovation index and innovation performance is stronger for large-sized firms and weaker for small-sized firms.

4. Methodology

4.1. Sample

To test the validity of our hypotheses, we analyzed the innovation-performance relationship for 886 listed firms across five countries: the USA, Germany, France, Canada, and the UK. These countries are among the world top ten economies in 2022, and are ranked 3rd, 4th, 10th, 11th, and 16th respectively out of 132 economies in the Global Innovation Index. France is the third-largest economic partner of the United Kingdom after the United States and Germany, and these three countries are the most important in Western Europe. The United States is the world's largest economy and Canada is a preferred investment destination. The dataset used in this study contains 16,606 observations extracted from the DataStream database.

| Country | Industry Sector | Number of firms | |
|----------------|----------------------------------|-----------------|-----|
| | Industrials | 24 | |
| Canada | Commercials | 10 | 4.4 |
| Canaua | Services | 9 | 44 |
| | Technology and Telecommunication | 1 | |
| | Industrials | 138 | |
| Franca | Commercials | 26 | 254 |
| France | Services | 58 | 234 |
| | Technology and Telecommunication | 32 | |
| | Industrials | 16 | |
| Cormony | Commercials | 2 | 25 |
| Germany | Services | 3 | 23 |
| | Technology and Telecommunication | 4 | |
| | Industrials | 107 | |
| United Kingdom | Commercials | 33 | 206 |
| United Kingdom | Services | 35 | 200 |
| | Technology and Telecommunication | 31 | |
| | Industrials | 230 | |
| United States | Commercials | 24 | 357 |
| Unicu States | Services | 14 | 337 |
| | Technology and Telecommunication | 89 | |

| Table L. Sample Firm | Selection by | v Country | and Industry | v Sector |
|----------------------|--------------|-----------|--------------|----------|
| | Selection of | , | | |

4.2. Dependent Variable

We evaluate a firm's financial performance based on its sales and profit volume (S&P), which reflects the actual level of profitability (Chege et al., 2020). Additionally, we incorporate Research and Development (R&D) expenditure as a measure of innovation performance (Usai et al., 2021), calculated by dividing total R&D expenditure by total assets.

4.3. Independent and Control Variables

To analyze the impact of Information and Communication Technology (ICT), we use data from the World Telecommunication/ICT Indicators database (June 2020 Edition). We consider two measures: the number of fixed-broadband subscriptions (FBS) per 100 inhabitants, which includes subscriptions to high-speed internet access via cable modem, DSL, fiber-to-the-home/building, satellite broadband, and terrestrial fixed wireless broadband; and internet users (IU), defined as the proportion of individuals who use the internet from any location. The data were derived from national household surveys. The estimated rate should reflect the total population of the country or, at least, individuals aged 5 years and older. We also incorporate the global index of innovation (GII) to measure a firm's innovation level. To ensure the reliability of the measurements, we reviewed relevant literature on entrepreneurship, innovation, and information systems. We consider two time periods: the crisis period (2007–2009) and the pre- and post-crisis (2000-2007 and 2010-2020) to understand the effects of the financial crisis on firm innovation performance. We include a year dummy variable to account for the financial crisis (with a value of 1 in 2007-2009 and 0 otherwise) and the COVID-19 crisis (with a value of 1 in 2019 and 2020 and 0 otherwise) on innovation performance.

To analyze the impact of country-specific institutional conditions, we include regulatory quality (RQ) and rule of law (RoL) (Marszk and Lechman, 2021) from worldwide governance indicators. We also control for macroeconomic conditions by using GDP growth from World Development Indicators. We control for firm age (Age) as a firm's experience-year since establishment and size (with a dummy variable based on the logarithm of total assets) to account for their effect on performance. To improve the robustness of our analysis, we perform additional tests using subsamples, classifying firms with size values above 15.326 (the median of the Size data is presented in *Table 2*) as large firms and small firms otherwise. Therefore, many studies focus on determining if small firms are better placed to capture the advantages of innovation on firm performance (Leiblein and Madsen, 2009; Rosenbusch, Brinckmann, and Bausch, 2011). We also conduct subsample tests to analyze the heterogeneity of the effects of the pandemic and the financial crisis in different countries.

4.4. Regression Models

We establish the following empirical models to examine the impact of the access to and the use of ICT on firms' performance and innovation performance.

$$FIRM _PERF_{i,t} = \beta_0 + \beta_1 GII_{i,t} + \beta_2 Fin - crisis_{i,t} + \beta_3 Covid - crisis_{i,t} + \beta_{4-10} Controls_{i,t} + \varepsilon_{i,t}$$
(1)

$$FIRM _PERF_{i,t} = \beta_0 + \beta_1 IU_{i,t} + \beta_2 Fin - crisis_{i,t} + \beta_3 Covid - crisis_{i,t} + \beta_{4-10} Controls_{i,t} + \varepsilon_{i,t}$$
(2)

$$FIRM _PERF_{i,t} = \beta_0 + \beta_1 FBS_{i,t} + \beta_2 Fin - crisis_{i,t} + \beta_3 Covid - crisis_{i,t} + \beta_{4-10} Controls_{i,t} + \varepsilon_{i,t}$$
(3)

$$INNOV _PERF_{i,t} = \beta_0 + \beta_1 GII_{i,t} + \beta_2 Fin - crisis_{i,t} + \beta_3 Covid - crisis_{i,t} + \beta_{4-10} Controls_{i,t} + \varepsilon_{i,t}$$
(4)

$$INNOV _PERF_{i,t} = \beta_0 + \beta_1 IU_{i,t} + \beta_2 Fin - crisis_{i,t} + \beta_3 Covid - crisis_{i,t} + \beta_{4-10} Controls_{i,t} + \varepsilon_{i,t}$$
(5)

$$INNOV _PERF_{i,t} = \beta_0 + \beta_1 FBS_{i,t} + \beta_2 Fin - crisis_{i,t} + \beta_3 Covid - crisis_{i,t} + \beta_{4-10} Controls_{i,t} + \varepsilon_{i,t}$$
(5)

$$INNOV _PERF_{i,t} = \beta_0 + \beta_1 FBS_{i,t} + \beta_2 Fin - crisis_{i,t} + \beta_3 Covid - crisis_{i,t} + \beta_{4-10} Controls_{i,t} + \varepsilon_{i,t}$$
(6)

In the above equation, i represents the firm and t represents the year. Our empirical analysis includes variables such as the Global Innovation Index $(_{GII_{it}})$ as a measure of innovation, Internet users $(_{IU_{it}})$, and Fixed

Broadband Subscriptions (FBS_{it}) in consecutive i-countries and t-years.

| Table II. ICT Sul | iiiiai y Stat | issues for the S | | Ties (2000–2020) Allilual Da | la |
|-------------------|---------------|------------------|---------------|------------------------------|-------------------------|
| Country | Obs | Mean | Std. dev | ICT penetration in 2000 | ICT penetration in 2020 |
| Internet Users (I | <u>U)</u> | | | | |
| USA | 21 | 71.95 | 12.16 | 43.08 | 90.9 |
| Germany | 21 | 73.40 | 17.31 | 30.22 | 89.81 |
| France | 21 | 63.54 | 22.35 | 14.31 | 84.8 |
| Canada | 21 | 78.98 | 12.78 | 51.3 | 96.97 |
| UK | 21 | 77.03 | 18.73 | 26.82 | 94.82 |
| Fixed Broadband | l Subscrip | tions (FBS) [[| oer 100 inhab | .] | |
| USA | 21 | 23.74 | 10.40 | 2.51 | 36.61 |
| Germany | 21 | 26.23 | 14.35 | 0.32 | 43.22 |
| France | 21 | 28.37 | 15.77 | 0.33 | 46.92 |
| Canada | 21 | 28.11 | 10.74 | 4.61 | 41.93 |
| UK | 21 | 25.72 | 13.84 | 0.09 | 40.26 |

Table II. ICT Summary Statistics for the Selected Countries (2000–2020) Annual Data

The study's dependent variables are the Sales and Profit (S&P) index for firm performance and Research and Development (R&D) as a measure of innovation performance. We also incorporate the financial crisis (*Fin.crisis_{i,t}*) and pandemic crisis (*Covid.crisis_{i,t}*) as independent variables, while controlling for other

variables such as firm age, country governance indicators (Rule of Law and Regulatory Quality), and GDP growth. The normal error term is represented by $\boldsymbol{\varepsilon_{it}}$. To examine changes in ICT access and use in the selected

countries, we use annual time series data on fixed broadband networks and Internet user rates for the period 2000-2020. These indicators measure the progress of ICT proliferation in sample countries presented in Table 2 (and graphically in Figures 1, 2, 3, 4, and 5).



Figure. 1,2,3,4, & 5. ICT diffusion trajectories from 2000 to 2020

Note: we graphically present the distribution of the ICT increases during the same period in each country of our sample in figures 1,2,3,4, and 5 in above. Y-axis: the share of the population with access to ICT; X-axis: time variation; Solid line: Internet Users; & Dashed-line: Fixed Broadband

5. Results

5.1. Summary Statistics

Table 3 presents the summary statistics of the final sample of 18,606 firm-year observations associated with 886 firms using mean, median, standard deviation, maximum, and minimum values. The average level of R&D expenditures for the full sample is \$11.77 million, with a large standard deviation of \$2.14 million, and a maximum value of \$17.57 million. As seen in Table 2, large firms tend to invest more in R&D (mean of 12.75 and median of 12.97) than small firms (mean of 0.42 and median of 10.55). Sales report an average \$ 14.70 million for the whole sample (and \$16.22 million when excluding small firms), indicating that large firms are more likely to have higher sales compared to small firms. The findings also reveal that the sales and investment in R&D are significantly lower for large firms than for small firms. The average firm's age in the sample is 37.72 years (ln-Age=3.63), with a standard deviation of 1.05 for the period from 2000 to 2020. The countries of our sample hold approximately 70% internet users, with a standard deviation of 0.13. The average fixed broadband subscriptions per 100 inhabitants is 26%, with a standard deviation of 0.13. The maximum value of subscriptions is 46.92, which indicates that the countries of our sample are unlikely to exceed half of fixed subscriptions for high-speed access to the public Internet. The results indicate that sales and R&D investment of large firms are significantly lower compared to small firms.

Table III. Descriptive Statistics

| | ample | | | | | |
|-----------|--------------|---------|--------|----------|--------|-------|
| Variables | Observations | Mean | Median | Std. dev | Min | Max |
| R&D | 6920 | 11.77 | 11.81 | 2.14 | 2.48 | 17.57 |
| S&P | 16426 | 14.70 | 14.97 | 2.28 | -1.24 | 20.14 |
| GII | 14407 | 0.33 | 0.54 | 0.29 | 0 | .624 |
| FBS | 18606 | 0.26 | 0.29 | 0.13 | 0.09 | 46.92 |
| IU | 18606 | 0.71 | 0.74 | 0.18 | 14.31 | 96.97 |
| Age | 17870 | 3.63 | 3.69 | 1.05 | 0 | 6.22 |
| GDP | 18606 | 0.01 | 0.02 | 0.02 | -0.09 | 0.07 |
| Size | 16395 | 15.03 | 15.33 | 2.27 | 0.51 | 20.5 |
| RQ | 17720 | 1.54 | 1.51 | 0.17 | 1.24 | 1.89 |
| RoL | 17720 | 1.49 | 1.56 | 0.20 | 1.07 | 1.88 |
| | e Dummy = 1 | | | | | |
| R&D | 4024 | 12.75 | 12.97 | 1.87 | 5.3 | 17.57 |
| S&P | 8338 | 16.2291 | 16.24 | 1.47 | -1.24 | 20.14 |
| GII | 4905 | 0.59 | 0.60 | 0.03 | .522 | 0.62 |
| FBS | 10408 | 25.93 | 29.46 | 12.62 | 0.09 | 46.92 |
| IU | 10408 | 71.86 | 74.55 | 17.43 | 14.31 | 96.97 |
| Age | 9765 | 3.76 | 3.93 | 1.06 | 0 | 6.22 |
| GDP | 10408 | 0.015 | 0.021 | 0.02 | -0.094 | 0.069 |
| RQ | 9935 | 1.53 | 1.5 | 0.174 | 1.24 | 1.89 |
| RoL | 9935 | 1.5 | 1.56 | 0.18 | 1.07 | 1.88 |
| | e Dummy = 0 | | | | | |
| R&D | 2896 | 10.42 | 10.55 | 1.71 | 2.48 | 14.61 |
| S&P | 8088 | 13.13 | 13.55 | 1.87 | 1.38 | 17.89 |
| GII | 3309 | 0.58 | 0.60 | 0.03 | .522 | 0.62 |
| FBS | 8198 | 25.66 | 28.66 | 13.96 | 0.09 | 46.92 |
| IU | 8198 | 70.16 | 74.55 | 18.95 | 14.31 | 96.97 |
| Age | 8105 | 3.47 | 3.43 | 1.003 | 0 | 6.21 |
| GDP | 8198 | 0.014 | 0.019 | 0.024 | -0.09 | 0.069 |
| RQ | 7785 | 1.56 | 1.51 | 0.168 | 1.24 | 1.89 |
| RoL | 7785 | 1.48 | 1.56 | 0.225 | 1.069 | 1.88 |

5.2. Correlation Results

Table 4 presents limited high levels of correlation between independent variables, which we intend to use separately in a regression model. Besides, the financial crisis is positively correlated with ICT variables and negatively correlated with the global innovation index (-0.4182). Importantly, both the global innovation index and ICTs infrastructure are significantly positively correlated with sales and R&D investment. Furthermore, there is a strong positive correlation between the levels of innovation measures and the pandemic crisis.

| Table | IV. Pears | on Correla | tion Matri | х | | | | | | | |
|------------|----------------------|---------------------|----------------------|-----------|-----------|-----------|-----------|--------------------|-----------|------------|----------|
| | R&D | GII | FBS | IU | Age | S&P | GDP | RoL | RQ | Fin.crisis | Covid-19 |
| R&D | 1.0000 | | | | | | | | | | |
| GII | 0.1622 ª | 1.0000 | | | | | | | | | |
| FBS | 0.0973 ^a | 0.7918 ^a | 1.0000 | | | | | | | | |
| IU | 0.0501 ª | 0.6931 ^a | 0.8747 ^a | 1.0000 | | | | | | | |
| Age | 0.0318 ^a | 0.1049 ^a | 0.1208 ^a | 0.1566 ª | 1.0000 | | | | | | |
| S&P | 0.5199ª | 0.0640 ª | 0.0408 ^a | 0.1445 ª | 0.3515 ª | 1.0000 | | | | | |
| GDP | 0.0007 | -0.1405 ª | -0.3390 ª | -0.2830 ª | -0.0367 ª | 0.0390ª | 1.0000 | | | | |
| RoL | -0.1734 ª | -0.3086 ª | -0.1938 ^a | 0.1674 ª | 0.0798 ª | 0.1993 ª | 0.217ª | 1.0000 | | | |
| RQ | -0.3181 ª | -0.3485 ª | -0.1946 ª | 0.0875 ª | -0.0171** | -0.0499 ª | 0.291* | 0.516ª | 1.0000 | | |
| Fin.crisis | -0.0256 ^b | -0.4182 ª | 0.0333 ª | 0.0610 ª | -0.0233 ª | -0.0066 | -0.297 ª | 0.079 ^a | 0.0525 ª | 1.0000 | |
| Covid-19 | 0.0770 ^a | 0.3258 ª | 0.3530ª | 0.3294 ª | 0.0799 ª | 0.0511 ª | -0.0511 ª | -0.220ª | -0.2635 ª | -0.1325 ª | 1.0000 |

^a and ^b present the level of significance of, respectively, 1% and 5%

5.3. Empirical Results

5.3.1. ICT, GII, and Firm Performance

First, we use the whole sample for the baseline regression analysis using the fixed-effect method. As economic changes occur, researchers and managers have attached more importance to determining the effect of ICT and GII on firm performance.

Table 5 reports the results from Eq. (1,2,3), where the dependent variable is sales revenue (sales). Regarding both GII and ICT penetration among internet users and fixed broadband subscriptions, we find significant positive effects on firm performance at the 1% level in all regressions. This result is consistent with earlier studies on firm performance, such as (De Guimarães et al., 2016; Centobelli, Cerchione, and Singh, 2019), who pointed out that innovation increases sales, market share and profits, eventually leading to an improvement in firm performance. The literature also highlights the importance of policy-makers taking an interest in the use of ICT to enhance firm performance (Gërguri-Rashiti et al., 2017).

Based on Table 5, it is evident that firm age (Age) has a significant and positive impact on firm performance for all models. This contradicts the findings of Ramadani et al., (2019) who suggest that firm age has a negative impact on firm performance in transition economies. The results indicate that the crisis coefficient is negative and significant at the 1% level, implying that firm performance is lower in the crisis period. This may be due to a lack of liquidity, insolvency problems, and government-related issues. The negative correlation between the financial crisis and firm performance suggests that the latter deteriorates during a global crisis (Tan, 2012). Finally, the GII and ICT coefficients remain statistically significant at the 1% level. After controlling for the pandemic crisis, they show weak coefficients (0.130, 0.156, 0.155 respectively) with a positive effect on firm performance. Despite the COVID-19 pandemic, firms have maintained their productivity levels and efficiency with a focus on the significance of healthcare systems, government policy, institutional systems, and good governance (Hu and Zhang, 2021). These results have significant political implications, and policymakers should ensure a favorable environment that emphasizes firm incentives to improve its performance.

| Models (FE) | S&P | S&P | S&P | S&P | S&P | S&P | S&P | S&P | S&P |
|--------------|-----------|-----------|-----------|-----------|------------|------------|-----------|-----------|-----------|
| GII | 0.579*** | | | 0.722*** | | | 0.562*** | | |
| | (0.0289) | | | (0.0332) | | | (0.0283) | | |
| FBS | | 1.952*** | | | 1.957*** | | | 1.926*** | |
| | | (0.0751) | | | (0.0750) | | | (0.0747) | |
| IU | | | 1.255*** | | | 1.261*** | | | 1.233*** |
| | | | (0.0571) | | | (0.0570) | | | (0.0570) |
| GDP | -0.697*** | 1.073*** | 1.252*** | 0.564* | 0.760*** | 0.895*** | -0.169 | 1.637*** | 1.806*** |
| | (0.259) | (0.240) | (0.242) | (0.288) | (0.283) | (0.284) | (0.268) | (0.247) | (0.248) |
| RoL | 0.144* | -0.00207 | 0.175*** | -0.218*** | 0.0455 | 0.231*** | 0.433*** | 0.217*** | 0.392*** |
| | (0.0743) | (0.0625) | (0.0639) | (0.0789) | (0.0635) | (0.0649) | (0.0867) | (0.0680) | (0.0693) |
| RQ | -0.232*** | -0.407*** | -0.683*** | -0.180*** | -0.383*** | -0.656*** | -0.218*** | -0.351*** | -0.623*** |
| | (0.0494) | (0.0444) | (0.0448) | (0.0490) | (0.0458) | (0.0461) | (0.0497) | (0.0452) | (0.0454) |
| Age | 1.078*** | 0.754*** | 0.854*** | 1.043*** | 0.745*** | 0.844*** | 1.058*** | 0.735*** | 0.836*** |
| | (0.0567) | (0.0456) | (0.0445) | (0.0570) | (0.0458) | (0.0448) | (0.0574) | (0.0459) | (0.0447) |
| Fin.crisis | | | | 0.225*** | -0.0418*** | -0.0482*** | | | |
| | | | | (0.0200) | (0.0145) | (0.0144) | | | |
| Covid-19 | | | | | | | 0.130*** | 0.156*** | 0.155*** |
| | | | | | | | (0.0218) | (0.0221) | (0.0222) |
| Constant | 10.82*** | 12.01*** | 11.41*** | 11.33*** | 11.94*** | 11.33*** | 10.41*** | 11.65*** | 11.05*** |
| | (0.239) | (0.198) | (0.193) | (0.250) | (0.197) | (0.192) | (0.231) | (0.196) | (0.190) |
| R-squared | 0.945 | 0.942 | 0.941 | 0.945 | 0.943 | 0.941 | 0.945 | 0.943 | 0.942 |
| Observation | 12,532 | 15,609 | 15,609 | 12,532 | 15,609 | 15,609 | 12,532 | 15,609 | 15,609 |
| Number of ID | 880 | 883 | 883 | 880 | 883 | 883 | 880 | 883 | 883 |

| Table V. The | Determinants | of the Firms' | Performance | (2000 - 2020) |
|------------------|--------------|---------------|---------------|---------------|
| I able i i I lle | Determinanto | | 1 of formunee | (2000 2020) |

*, **, ***, present the level of significance of, respectively, 10%, 5%, and 1%.

5.3.2. Firm Size and Firm Performance

The subsample tests in Table 6 verify the positive effects of ICT applications on firm performance. High levels of ICT implementation and GII are associated with improved performance for both high and low-sized firms, especially during the pandemic. We find that the relationship between the financial crisis and firm performance differs in our estimations, as the measurement of innovation is different.

| Table VI. The Effect of Firm Size on the Relationsh | ip between GII, ICT, and Firm Performance |
|---|---|
|---|---|

| Models (FE) | S&P | S&P | S&P | S&P | S&P | S&P |
|--------------|----------|-----------|-----------|---------------|-----------|-----------|
| | | | | Low Sized Fir | ms | |
| GII | 0.534*** | | | 0.662*** | | |
| | (0.0344) | | | (0.0450) | | |
| FBS | | 1.592*** | | | 1.783*** | |
| | | (0.0875) | | | (0.113) | |
| IU | | | 1.081*** | | | 1.049*** |
| | | | (0.0669) | | | (0.0837) |
| GDP | 1.568*** | 1.540*** | 1.620*** | 1.485*** | 2.086*** | 2.171*** |
| | (0.331) | (0.292) | (0.292) | (0.393) | (0.429) | (0.428) |
| RoL | 0.0700 | 0.0535 | 0.205*** | 0.101 | 0.416*** | 0.556*** |
| | (0.0850) | (0.0678) | (0.0699) | (0.121) | (0.0955) | (0.0979) |
| RQ | -0.0288 | -0.0703* | -0.357*** | -0.322*** | -0.709*** | -0.814*** |
| | (0.0427) | (0.0394) | (0.0409) | (0.0858) | (0.0812) | (0.0821) |
| Age | 0.428*** | 0.301*** | 0.360*** | 1.360*** | 0.866*** | 0.999*** |
| | (0.0715) | (0.0614) | (0.0612) | (0.0773) | (0.0691) | (0.0656) |
| Fin.crisis | 0.177*** | -0.0222 | -0.0296* | 0.223*** | 0.0185 | 0.0163 |
| | (0.0271) | (0.0166) | (0.0164) | (0.0234) | (0.0211) | (0.0211) |
| Covid-19 | 0.131*** | 0.0966*** | 0.0636*** | 0.171*** | 0.198*** | 0.237*** |
| | (0.0209) | (0.0201) | (0.0199) | (0.0326) | (0.0375) | (0.0375) |
| Constant | 14.25*** | 14.60*** | 14.21*** | 8.580*** | 10.08*** | 9.295*** |
| | (0.283) | (0.234) | (0.229) | (0.327) | (0.285) | (0.268) |
| R-squared | 0.937 | 0.934 | 0.933 | 0.928 | 0.914 | 0.912 |
| Observation | 6,796 | 8,008 | 8,008 | 5,736 | 7,601 | 7,601 |
| Number of ID | 584 | 593 | 593 | 551 | 599 | 599 |

*, **, ***, present the level of significance of, respectively, 10%, 5%, and 1%.

5.3.3. ICT, GII, and Innovation-Performance

To test the effect of GII and the use of ICT on a firm's innovation performance, we used the whole sample for the baseline regression analysis conducted using both fixed and random effect method. The estimated results are presented in *Table 7*. Our results reveal that firms can enhance their innovation performance significantly by utilizing ICTs in their process. The variables show statistically positive coefficients in all estimations, indicating that firms in countries with higher ICT implementation are more likely to invest in R&D, which leads to an improvement in their innovation performance. By examining the coefficients of GII and ICT, we can confirm that access and use of ICT (the use of internet and fixed broadband subscriptions) are more likely associated with innovation performance than GII (global innovation index). Our results align with those of previous studies by Ren-Jye, Hsien, and Keisuke, (2018); Wang and Qi, (2021); Ueki and Tsuji, (2019). In particular, Ueki and Tsuji (2019) find that ICTs infrastructure has a good promoting effect on innovation performance. They underline that ICTs allow organizations to acquire external information in new technology and to share internal knowledge about product, production, market, and consumers.

The results remain robust even after controlling for the crises impact. In fact, the COVID-19 pandemic has shown a positive effect on innovation performance, as demonstrated by the study of Han and Qian (2021), who found that the innovative capabilities of Chinese listed firms increased during the pandemic. We suggest that maintaining strong internal and external knowledge capabilities can help firms to mitigate the effects of the crisis. Firms located in countries with good healthcare systems and good governance are more likely to invest in R&D and acquire external knowledge. However, the financial crisis affects differently innovation performance. It generally has a negative and significant effect. Tan (2012) argues that crises reinforce the negative relationship between financial distress and firm performance, such that highly leveraged firms experience the worst performance during crises.

| Models | R&D (RE) | R&D (RE) | R&D (RE) | R&D (RE) | R&D (RE) | R&D (RE) | R&D (FE) | R&D (FE) | R&D (FE) |
|-----------------|-----------|-------------|-------------|----------|------------|------------|-----------|-------------|-----------|
| GII | 0.726*** | ~ / | ~ / | 0.857*** | | | 0.671*** | | |
| | (0.0681) | | | (0.0751) | | | (0.0347) | | |
| FBS | · / | 1.806*** | | | 1.805*** | | · / | 1.650*** | |
| | | (0.235) | | | (0.235) | | | (0.0922) | |
| IU | | | 1.268*** | | | 1.269*** | | · / | 1.132*** |
| | | | (0.161) | | | (0.161) | | | (0.0707) |
| GDP | -1.732*** | 1.160*** | 1.472*** | -0.0500 | 0.530 | 0.817** | -1.354*** | 1.443*** | 1.638*** |
| | (0.420) | (0.300) | (0.303) | (0.410) | (0.358) | (0.351) | (0.417) | (0.391) | (0.395) |
| RoL | 0.176 | -0.0113 | 0.268** | -0.186 | 0.0792 | 0.362*** | 0.488*** | 0.331*** | 0.567*** |
| | (0.137) | (0.141) | (0.131) | (0.125) | (0.127) | (0.121) | (0.132) | (0.117) | (0.116) |
| RQ | -0.243*** | -0.590*** | -0.900*** | -0.209** | -0.557*** | -0.866*** | -0.170** | -0.490*** | -0.774*** |
| | (0.0921) | (0.106) | (0.102) | (0.0913) | (0.107) | (0.102) | (0.0698) | (0.0674) | (0.0663) |
| Age | 0.748*** | 0.560*** | 0.608*** | 0.722*** | 0.550*** | 0.597*** | 0.833*** | 0.628*** | 0.688*** |
| - | (0.142) | (0.142) | (0.138) | (0.142) | (0.142) | (0.139) | (0.0359) | (0.0359) | (0.0353) |
| Fin.crisis | | | | 0.214*** | -0.0676*** | -0.0706*** | | | |
| | | | | (0.0256) | (0.0260) | (0.0258) | | | |
| Covid-19 | | | | | | | 0.117*** | 0.159*** | 0.143*** |
| | | | | | | | (0.0357) | (0.0344) | (0.0348) |
| Constant | 8.595*** | 9.821*** | 9.223*** | 9.105*** | 9.691*** | 9.087*** | 7.839*** | 9.120*** | 8.552*** |
| | (0.584) | (0.591) | (0.567) | (0.597) | (0.579) | (0.558) | (0.260) | (0.243) | (0.239) |
| R-squared | - | - | - | - | - | - | 0.941 | 0.932 | 0.931 |
| Chi2 | 289.45 | 246.15 | 255.22 | 325.96 | 248.98 | 258.39 | - | - | - |
| Prob > chi2 | 0.0000 | 0.0000 | 0.0000 | 0.0000 | 0.0000 | 0.0000 | - | - | - |
| Observation | 5,790 | 6,592 | 6,592 | 5,790 | 6,592 | 6,592 | 5,790 | 6,592 | 6,592 |
| Number of ID | 392 | 407 | 407 | 392 | 407 | 407 | 392 | 407 | 407 |

| Table VII. The Determinants of the Finn S innovation-renormance $(2000-2020)$ |
|--|
|--|

*, **, ***, present the level of significance of, respectively, 10%, 5%, and 1%.

5.3.4. Firm Size and Innovation-Performance

The subsample tests reported in Table 8 confirm the positive effects of the COVID-19 pandemic. Specifically, for small-sized firms, their innovation capabilities increase significantly during the crisis. Faced with the pandemic, firms are more likely to invest in research and development to seek new opportunities to survive and make profits. The crisis represents an opportunity for small firms to innovate in order to respond to new demands. Policy-makers elaborate strategies that support small firms, especially through subsidies, to ensure their continuity. Small firms are more vulnerable and likely to disappear during crises. In columns (2) and (3), the financial crisis primarily affected large firms and may have contributed to the fact that highly leveraged firms have lower performance, as documented by Tan (2012). This, in turn, negatively affects firms' decisions regarding R&D investments.

| Models | R&D (RE) | R&D (RE) R&D (RE) | | R&D (FE) R&D (FE) | | R&D (FE) | | | |
|--------------|----------------|-------------------|-----------|-------------------|-----------|-----------|--|--|--|
| | High Sized fir | ms | | Low Sized Firms | | | | | |
| GII | 0.660*** | | | 0.691*** | | | | | |
| | (0.0832) | | | (0.0667) | | | | | |
| FBS | | 1.612*** | | | 1.259*** | | | | |
| | | (0.256) | | | (0.182) | | | | |
| IU | | | 1.159*** | | | 0.797*** | | | |
| | | | (0.168) | | | (0.138) | | | |
| GDP | 0.387 | 0.506 | 0.689* | 0.328 | 1.206 | 1.255 | | | |
| | (0.440) | (0.374) | (0.364) | (0.775) | (0.791) | (0.793) | | | |
| RoL | 0.00231 | 0.0741 | 0.270 | 0.238 | 0.639*** | 0.821*** | | | |
| | (0.203) | (0.204) | (0.211) | (0.195) | (0.188) | (0.186) | | | |
| RQ | 0.0380 | -0.143 | -0.459*** | -0.172 | -0.525*** | -0.659*** | | | |
| | (0.0961) | (0.104) | (0.107) | (0.114) | (0.116) | (0.114) | | | |
| Age | 0.171 | -0.000150 | 0.0370 | 1.233*** | 1.117*** | 1.203*** | | | |
| | (0.120) | (0.129) | (0.127) | (0.100) | (0.104) | (0.101) | | | |
| Fin.crisis | 0.166*** | -0.0723** | -0.0809** | 0.158*** | -0.0192 | -0.0172 | | | |
| | (0.0374) | (0.0319) | (0.0317) | (0.0402) | (0.0340) | (0.0339) | | | |
| Covid-19 | 0.0918* | 0.0711 | 0.0293 | 0.232*** | 0.275*** | 0.288*** | | | |
| | (0.0498) | (0.0590) | (0.0595) | (0.0653) | (0.0727) | (0.0723) | | | |
| Constant | 11.35*** | 11.97*** | 11.58*** | 5.554*** | 5.786*** | 5.125*** | | | |
| | (0.552) | (0.567) | (0.575) | (0.450) | (0.481) | (0.456) | | | |
| R-squared | - | - | - | 0.930 | 0.913 | 0.912 | | | |
| Chi2 | 83.65 | 61.52 | 68.13 | - | - | - | | | |
| Prob > chi2 | 0.0000 | 0.0000 | 0.0000 | - | - | - | | | |
| Observation | 3,390 | 3,860 | 3,860 | 2,400 | 2,732 | 2,732 | | | |
| Number of ID | 297 | 304 | 304 | 297 | 304 | 304 | | | |

*, **, ***, present the level of significance of, respectively, 10%, 5%, and 1%.

5.4. Additional Analysis: Countries Specific Analysis

To strengthen the credibility of our findings, we conducted several robustness checks. Specifically, we conducted a cross-country analysis to examine whether the innovation-performance relationship varies under different country conditions.

Table 9 underlines the strong positive relationship between internet users and innovation performance, which is highly dependent on the proportion of internet users. Canada stands out with the highest proportion of internet users at 96.97%, leading to greater innovation performance. On the other hand, in the French context, internet users are less likely to be associated with innovation performance. Regulatory quality is an important factor contributing to economic activities and firm performance, but its relationship with innovation performance varies across countries. In line with Arvate, Barbosa, and Fuzitani (2013), higher regulatory quality in the USA may lead to lower firm performance and lower investment in research and development. In contrast, Frotagheh and Kardan (2019) find a positive and significant relationship between regulatory quality and innovation performance in the UK. The legal environment also plays a role in business activities. Haggard and Tiede (2011) highlight that the rule of law is strongly associated with institutional development and economic growth, which provide a favorable environment for firms to make business. Our findings in the American context are consistent with previous studies (Gomez, 2016).

However, we find a negative and significant relationship between the rule of law and firm performance in the UK and Germany, according to Frotagheh and Kardan (2019).

Based on our estimation that firm performance deteriorates during a crisis, we can confirm that the global financial crisis had a negative impact on both firm and innovation performance across the entire sample, with the greatest effects observed in the USA and Germany. This result supports the idea that the financial crisis has made

business opportunities less certain, therefore forcing firms to postpone long-term innovation investments (Archibugi, Filippetti, and Frenz, 2013; Colombo et al., 2016; Filippetti and Archibugi, 2011; Paunov, 2012). Moreover, innovation is now considered a crucial strategic response to crises such as the COVID-19 pandemic (Zhong et al., 2022). Our empirical results indicate that innovation performance is positively associated with the pandemic crisis in both the French and British contexts.

Table 10 indicates the relationship between firm performance, innovation performance, and fixed broadband subscriptions. Our findings suggest that the relationship between fixed broadband infrastructure and innovation performance is strong in Canada (β =4.530) and weak in France (β =1.476), despite the fact that France required more infrastructure than the other countries in our sample. The global financial crisis had a negative effect on innovation performance in American, German, and British firms, possibly due to the close relationships between these countries and their partners. Similar to the results presented in *Table 9*, our empirical findings indicate that the pandemic crisis has a similar effect on innovation performance.

6. Conclusion

Using fixed and random effects, this study examines the impact of ICTs proliferation on firms-performance and innovation-performance taking into account the financial and pandemic crisis. Additionally, we classify firms by their stock market indices and conduct a subsample test. Our results demonstrate that high levels of ICT access and use are associated with both firm and innovation performance (measured by R&D) for both small and large companies. Moreover, large firms tend to be strong innovators and invest more in R&D.

However, we also find that the COVID-19 pandemic has significantly increased R&D investment among small companies, and its impact varies across countries. Similarly, the impact of the financial crisis on performance depends on the measurement index of innovation, with large firms being the most vulnerable. Furthermore, our cross-country analysis reveals heterogeneous results. Our findings suggest a significant relationship between regulatory quality, rule of law and innovation performance, which can be direct or inverse depending on the measurement index of ICT implementation. This may be due to changes in governance across countries and governments.

As with any research, this study has certain limitations that could provide directions for future research. Firstly, the geographical scope of the sample is limited to five developed countries, and it would be beneficial to extend it to other regions and countries to conduct a comparative study of innovation behavior between more developed and emergent countries. Another limitation is related to innovation measures, and future research should consider using other innovation indices to measure their effect on firm performance. Additionally, missing data on R&D expenditure limits our analysis, and the ICT variables do not fully capture the range of technologies available to companies, such as data analytics, big data, and artificial intelligence.

| Table | e IX. | Firm | Performance | and | Innovation-Performance | Versus | Internet | Users: | Country-Specific |
|-------|---------|--------|-------------|-----|------------------------|--------|----------|--------|------------------|
| Analy | ysis (2 | 000-20 | 20) | | | | | | |

| | \ | / | 1 | | 1 | | 1 | | | |
|-------------|-----------|------------|-----------|------------|----------|----------|-----------|----------|-----------|------------|
| Country | USA | | Germany | | France | | Canada | | UK | |
| Perf/Innov- | Firm-Perf | Innov-Perf | Firm-Perf | Innov-Perf | Firm- | Innov- | Firm-Perf | Innov- | Firm-Perf | Innov-Perf |
| Perf | S&P(FE) | R&D(FE) | S&P(RE) | R&D(RE) | Perf | Perf | S&P(RE) | Perf | S&P(RE) | R&D(RE) |
| | | | | | S&P(FE) | R&D(RE) | | R&D(RE) | | |
| IU | 1.920*** | 1.944*** | 1.462*** | 1.205** | 0.908*** | 0.477* | 3.746*** | 4.243*** | 1.648*** | 1.730*** |
| | (0.123) | (0.139) | (0.314) | (0.575) | (0.108) | (0.278) | (0.564) | (1.345) | (0.194) | (0.269) |
| GDP | 1.600*** | 0.483 | 0.0168 | -2.979 | 2.541*** | 0.439 | 0.150 | 5.545*** | -0.0112 | -1.289*** |
| | (0.573) | (0.728) | (0.664) | (1.942) | (0.657) | (0.969) | (0.706) | (2.047) | (0.328) | (0.486) |
| RoL | 0.577*** | 0.687*** | -0.598* | -1.810* | 0.0727 | 0.428* | -0.357 | 0.0964 | -0.595*** | -0.362 |
| | (0.154) | (0.185) | (0.320) | (0.937) | (0.144) | (0.243) | (0.360) | (1.742) | (0.120) | (0.251) |
| RQ | -0.529*** | -0.556*** | -0.0815 | 0.893 | -0.404 | -0.531* | 0.173 | 0.137 | 0.885*** | 0.902*** |
| | (0.0504) | (0.0614) | (0.173) | (1.061) | (0.299) | (0.316) | (0.253) | (1.827) | (0.160) | (0.268) |
| Age | 1.061*** | 0.935*** | 0.0304 | -0.267 | 0.611*** | -0.348 | 0.772*** | 0.864 | 0.599*** | 0.0919 |
| | (0.0869) | (0.101) | (0.166) | (0.321) | (0.0768) | (0.316) | (0.264) | (0.530) | (0.151) | (0.211) |
| Fin.crisis | -0.0582** | -0.114*** | -0.144*** | -0.346*** | 0.00937 | 0.0823 | -0.0395 | -0.0958 | -0.0433 | -0.120** |
| | (0.0257) | (0.0303) | (0.0556) | (0.0992) | (0.0335) | (0.0619) | (0.0592) | (0.223) | (0.0368) | (0.0540) |
| Covid-19 | -0.00484 | 0.0176 | 0.0955 | -0.105 | 0.226*** | -0.415 | 0.0113 | -0.389 | 0.245*** | 0.246*** |
| | (0.0434) | (0.0550) | (0.0798) | (0.256) | (0.0490) | (0.275) | (0.0688) | (0.337) | (0.0350) | (0.0606) |
| Constant | 10.13*** | 7.147*** | 16.38*** | 14.07*** | 11.07*** | 12.52*** | 9.823*** | 4.117 | 9.817*** | 6.770*** |
| | (0.355) | (0.426) | (0.700) | (1.707) | (0.403) | (1.267) | (1.098) | (5.341) | (0.566) | (0.918) |
| R-squared | 0.894 | 0.920 | - | - | 0.943 | - | - | - | - | - |
| Chi2 | - | - | 49.71 | 20.73 | - | 24.95 | 131.74 | 59.77 | 272.94 | 71.60 |
| Prob > chi2 | - | - | 0.0000 | 0.0042 | - | 0.0008 | 0.0000 | 0.0000 | 0.0000 | 0.0000 |
| Observation | 6,746 | 3,966 | 488 | 448 | 4,104 | 692 | 782 | 121 | 3,489 | 1,365 |
| Number of | 357 | 229 | 25 | 24 | 252 | 51 | 44 | 13 | 205 | 90 |
| ID | | | | | | | | | | |

Performance is proxied by sales and profit "S&P" and Research and Development (R&D) as a measure of innovation-performance. ICT is proxied by two types of data from the World Telecommunication/ICT Indicators database (June 2020 Edition): the number of fixed-broadband subscriptions (FBS) per 100 inhabitants: that refers to fixed subscriptions to high-speed access to the public Internet via cable modem, DSL; fiber-to-thehome/building; other fixed (wired)-broadband subscriptions; satellite broadband and terrestrial fixed wireless broadband. Technically, it includes fixed (wired) broadband; satellite broadband, and terrestrial fixed wireless broadband. Besides, we use also Internet users (IU), defined as the "proportion of individuals who used the Internet from any location". The data were derived from national household surveys. The estimated rate should reflect the total population of the country or, at least individuals aged 5 years and older. We also incorporate the global index of innovation (GII) as a measure of a firm's innovation level. Regulatory Quality (RQ) and rule of law (RoL) from Worldwide Governance Indicators to detect the impact of country governance level on innovation-performance. GDP growth from World Development Indicators to control the macroeconomic conditions through countries. The firm age (Age) is a firm's experience-year since establishment. The size is a dummy variable to control the firm size effect (the logarithm of the total asset). We specify a year dummy to take account of the financial crisis that takes the value 1 in the years 2007–2009, and 0 otherwise (2000–2007 and 2010-2020). Also, we incorporate the Covid19 crisis (2019-2020). We specify a year dummy to take account of the Covid-19 crisis that takes the value 1 in the years 2019 and 2020, and 0 otherwise.

Standard errors in parentheses *, **, ***, present the level of significance of, respectively, 10%, 5%, and 1%.

 Table X. Firm Performance and Innovation-Performance Versus Fixed Broadband Subscriptions: Country-Specific Analysis (2000-2020)

| | Č | | | | | | | | | |
|-----------------|----------|-----------|----------|-----------|----------|----------|------------|-----------|------------|----------|
| Country | USA | | Germany | | France | | Canada | | UK | |
| Perf/Innov- | Innov- | Firm-Perf | Innov- | Firm- | Innov- | Firm- | Innov-Perf | Firm-Perf | Innov-Perf | Firm- |
| Perf | Perf | S&P(FE) | Perf | Perf | Perf | Perf | R&D(RE) | S&P(RE) | R&D(RE) | Perf |
| | R&D(FE) | | R&D(RE) | S&P(RE) | R&D(FE) | S&P(RE) | | | | S&P(RE) |
| FBS | 2.951*** | 2.885*** | 1.834*** | 1.536** | 1.476*** | 0.631 | 4.530*** | 5.140*** | 2.277*** | 2.430*** |
| | (0.166) | (0.192) | (0.379) | (0.650) | (0.157) | (0.462) | (0.695) | (1.536) | (0.250) | (0.366) |
| GDP | 2.194*** | 1.048 | 0.248 | -2.796 | 2.510*** | 0.0796 | -0.327 | 4.706** | 0.897*** | -0.203 |
| | (0.579) | (0.735) | (0.676) | (1.939) | (0.647) | (1.100) | (0.702) | (1.856) | (0.302) | (0.467) |
| RoL | -0.135 | 0.0158 | 0.346 | -1.033 | 0.141 | 0.396 | 0.281 | 0.814 | -0.458*** | -0.255 |
| | (0.156) | (0.190) | (0.324) | (1.014) | (0.142) | (0.248) | (0.339) | (1.741) | (0.111) | (0.239) |
| RQ | 0.110** | 0.0659 | 0.148 | 1.082 | -0.577** | -0.468 | 0.345 | 0.189 | 0.177 | 0.146 |
| | (0.0508) | (0.0656) | (0.166) | (1.059) | (0.289) | (0.382) | (0.273) | (1.916) | (0.137) | (0.244) |
| Age | 0.984*** | 0.876*** | -0.0389 | -0.322 | 0.560*** | -0.335 | 0.797*** | 0.902* | 0.521*** | 0.00775 |
| | (0.0879) | (0.101) | (0.161) | (0.318) | (0.0793) | (0.306) | (0.265) | (0.530) | (0.151) | (0.214) |
| Fin.crisis | -0.00646 | -0.0619** | -0.0722 | -0.288*** | 0.0664* | 0.102* | -0.143** | -0.201 | -0.0420 | -0.114** |
| | (0.0259) | (0.0300) | (0.0509) | (0.101) | (0.0344) | (0.0576) | (0.0641) | (0.208) | (0.0367) | (0.0539) |
| Covid-19 | 0.0558 | 0.0889* | 0.0867 | -0.117 | 0.175*** | -0.448* | 0.0645 | -0.322 | 0.147*** | 0.135** |
| | (0.0407) | (0.0522) | (0.0769) | (0.250) | (0.0487) | (0.267) | (0.0708) | (0.369) | (0.0326) | (0.0593) |
| Constant | 11.24*** | 8.186*** | 15.56*** | 13.38*** | 11.56*** | 12.54*** | 10.21*** | 4.796 | 11.78*** | 8.918*** |
| | (0.385) | (0.456) | (0.598) | (1.697) | (0.418) | (1.243) | (1.109) | (5.160) | (0.621) | (0.996) |
| R-squared | 0.896 | 0.921 | - | - | 0.943 | - | - | - | - | - |
| Chi2 | - | - | 51.87 | 21.03 | - | 22.14 | 131.44 | 53.39 | 293.39 | 74.35 |
| Prob > chi2 | - | - | 0.0000 | 0.0037 | - | 0.0024 | 0.0000 | 0.0000 | 0.0000 | 0.0000 |
| Observation | 6,746 | 3,966 | 488 | 448 | 4,104 | 692 | 782 | 121 | 3,489 | 1,365 |
| Number of ID | 357 | 229 | 25 | 24 | 252 | 51 | 44 | 13 | 205 | 90 |

Performance is proxied by sales and profit "S&P" and Research and Development (R&D) as a measure of innovation-performance. ICT is proxied by two types of data from the World Telecommunication/ICT Indicators database (June 2020 Edition): the number of fixed-broadband subscriptions (FBS) per 100 inhabitants: that refers to fixed subscriptions to high-speed access to the public Internet via cable modem, DSL; fiber-to-thehome/building; other fixed (wired)-broadband subscriptions; satellite broadband and terrestrial fixed wireless broadband. Technically, it includes fixed (wired) broadband; satellite broadband, and terrestrial fixed wireless broadband. Besides, we use also Internet users (IU), defined as the "proportion of individuals who used the Internet from any location". The data were derived from national household surveys. The estimated rate should reflect the total population of the country or, at least individuals aged 5 years and older. We also incorporate the global index of innovation (GII) as a measure of a firm's innovation level. Regulatory Quality (RQ) and rule of law (RoL) from Worldwide Governance Indicators to detect the impact of country governance level on innovation-performance. GDP growth from World Development Indicators to control the macroeconomic conditions through countries. The firm age (Age) is a firm's experience-year since establishment. The size is a dummy variable to control the firm size effect (the logarithm of the total asset). We specify a year dummy to take account of the financial crisis that takes the value 1 in the years 2007–2009, and 0 otherwise (2000–2007 and 2010-2020). Also, we incorporate the Covid19 crisis (2019-2020). We specify a year dummy to take account of the Covid-19 crisis that takes the value 1 in the years 2019 and 2020, and 0 otherwise.

Standard errors in parentheses *, **, ***, present the level of significance of, respectively, 10%, 5%, and 1%.

Acknowledgements

The authors report there are no competing interests to declare.

Declaration of Interest Statement

The authors report there is no conflict of interests to declare.

References

Agostini, Lara, Caviggioli, Federico, Filippini, Roberto, and Nosella, Anna, (2015). "Does patenting influence SME sales performance? A quantity and quality analysis of patents in Northern Italy", *European Journal of Innovation Management*, Vol. 18 No. 2, pp. 238-257. Doi: 10.1108/EJIM-07-2013-0071 Amankwah-Amoah, Joseph. (2021). "COVID-19 Pandemic and Innovation Activities in the Global Airline Industry: A Review." Environment International, Vol. 156, p. 106719. Doi: 10.1016/j.envint.2021.106719

- Ani, D. P., and M. Ofikwu. (2019). "Factors Affecting the Use of Information and Communication Technologies (ICTS) in Marketing of Grains among Women in Benue State." Doi: 10.22004/ag.econ.295742
- Aragón-Sánchez, Antonio, and Gregorio Sánchez-Marín. (2005). "Strategic Orientation, Management Characteristics, and Performance: A Study of Spanish SMEs." *Journal of Small Business Management, Vol.* 43 No.3, pp. 287–308. Doi: 10.1111/j.1540-627X.2005.00138.x
- Archibugi, Daniele. (2017). "Blade Runner Economics: Will Innovation Lead the Economic Recovery?" *Research Policy*, Vol. 46 No.3, pp. 535–543. Doi: 10.1016/j.respol.2016.01.021
- Archibugi, Daniele, and Alberto Coco. (2004). "A New Indicator of Technological Capabilities for Developed and Developing Countries (ArCo)." World Development, Vol. 32 No.4, pp. 629–654. Doi: 10.1016/j.worlddev.2003.10.008
- Archibugi, Daniele, Andrea Filippetti, and Marion Frenz. (2013). "The Impact of the Economic Crisis on Innovation: Evidence from Europe." *Technological Forecasting and Social Change*, Vol. 80 No.7, pp. 1247–1260. Doi: 10.1016/j.techfore.2013.05.005
- Arias-Aranda, Daniel, Beatriz Minguela-Rata, and Antonio Rodríguez-Duarte. (2001). "Innovation and Firm Size: An Empirical Study for Spanish Engineering Consulting Companies." *European Journal of Innovation Management*, Vol. 4 No. 3, pp. 133-142. Doi: 10.1108/EUM000000005671
- Arrow, Kenneth. (1962). "Economic Welfare and the Allocation of Resources for Invention." In *The Rate and Direction of Inventive Activity: Economic and Social Factors*, pp. 609–26. Princeton University Press. Available at: https://www.nber.org/system/files/chapters/c2144/c2144.pdf
- Arvanitis, Spyros, Euripidis Loukis, and Vasiliki Diamantopoulou. (2013). "The Effect of Soft ICT Capital on Innovation Performance of Greek Firms." *Journal of Enterprise Information Management*, Vol. 26 No. 6, pp. 679-701. Doi: 10.1108/JEIM-07-2013-0048
- Arvate, Paulo Roberto, Klenio Barbosa, and Eric Fuzitani. (2013). "Campaign Donation and Government Contracts in Brazilian States." Available at: http://hdl.handle.net/10438/11325
- Audretsch, David B., and Zoltan J. Acs. (1991). "Innovation and Size at the Firm Level." Southern Economic Journal, Vol. 57, No. 3, pp. 739–744. Doi: 10.2307/1059787
- Azar, Goudarz, and Francesco Ciabuschi. (2017). "Organizational Innovation, Technological Innovation, and Export Performance: The Effects of Innovation Radicalness and Extensiveness." *International Business Review*, Vol. 26 No. 2, pp. 324–336. Doi: 10.1016/j.ibusrev.2016.09.002
- Bankole, Felix Olu, Kweku-Muata Osei-Bryson, and Irwin Brown. (2015). "The Impacts of Telecommunications Infrastructure and Institutional Quality on Trade Efficiency in Africa." *Information Technology for Development*, Vol. 21 No. 1, pp. 29–43. Doi: 10.1080/02681102.2013.874324
- Barbieri, Elisa, Marco R. Di Tommaso, Chiara Pollio, and Lauretta Rubini. (2020). "Getting the Specialization Right. Industrialization in Southern China in a Sustainable Development Perspective." World Development, Vol. 126, p. 104701. Doi: 10.1016/j.worlddev.2019.104701
- Bardhan, Indranil, Viswanathan Krishnan, and Shu Lin. (2013). "Research Note—Business Value of Information Technology: Testing the Interaction Effect of IT and R&D on Tobin's Q." *Information Systems Research*, Vol. 24 No. 4, pp. 1147–1161. Doi: 10.1287/isre.2013.0481
- Baumann, Julian, and Alexander S. Kritikos. (2016). "The Link between R&D, Innovation and Productivity: Are Micro Firms Different?" *Research Policy*, Vol. 45 No. 6, pp. 1263–1274. Doi: 10.1016/j.respol.2016.03.008
- Biron, Michal, Helen De Cieri, Ingrid Fulmer, Cai-Hui Veronica Lin, Wolfgang Mayrhofer, Margarita Nyfoudi, Karin Sanders, Helen Shipton, and Jian Min James Sun. (2021). "Structuring for Innovative Responses to Human Resource Challenges: A Skunk Works Approach." *Human Resource Management Review*, Vol. 31 No. 2, p. 100768. Doi: 10.1016/j.hrmr.2020.100768
- Bloodgood, James M., Harry J. Sapienza, and James G. Almeida. (1996). "The Internationalization of New High-Potential US Ventures: Antecedents and Outcomes." *Entrepreneurship Theory and Practice*, Vol. 20 No. 4, pp. 61–76. Doi: 10.1177/104225879602000405
- Bodlaj, Mateja, Selma Kadic-Maglajlic, and Irena Vida. (2020). "Disentangling the Impact of Different Innovation Types, Financial Constraints and Geographic Diversification on SMEs' Export Growth." *Journal of Business Research*, Vol. 108, pp. 466–475. Doi: 10.1016/j.jbusres.2018.10.043
- Bollou, Felix. (2006). "ICT Infrastructure Expansion in Sub-Saharan Africa: An Analysis of Six West African Countries from 1995 to 2002." *The Electronic Journal of Information Systems in Developing Countries*, Vol. 26 No. 1, pp. 1–16. Doi: 10.1002/j.1681-4835.2006.tb00174.x
- Botello, Hector, and Aura Pedraza Avella. (2014). "Las Tecnologías De La Información Y La Comunicación YEl Desempeño De Las Firmas: Evidencia De Las Firmas Industriales Del Ecuador (The InformationTechnology and Communication and Firm Performance: Evidence from Firms Industrialists Ecuador)." InCPRLATAM-CommunicationPolicyResearchConference.Availableat:

SSRN: https://ssrn.com/abstract=2462537 or http://dx.doi.org/10.2139/ssrn.2462537

- Breier, Matthias, Andreas Kallmuenzer, Thomas Clauss, Johanna Gast, Sascha Kraus, and Victor Tiberius. (2021). "The Role of Business Model Innovation in the Hospitality Industry during the COVID-19 Crisis." *International Journal of Hospitality Management*, Vol. 92, p. 102723. Doi: 10.1016/j.ijhm.2020.102723
- Brynjolfsson, Erik, and Adam Saunders. (2010). "Wired for Innovation." *How Information Technology in Reshaping the Economy. Massachusetts Institute of Technology. USA.* Available at: https://core.ac.uk/download/pdf/82648065.pdf
- Cardona, Melisande, Tobias Kretschmer, and Thomas Strobel. (2013). "ICT and Productivity: Conclusions from the Empirical Literature." *Information Economics and Policy*, Vol. 25 No. 3), pp.109–125. Doi: 10.1016/j.infoecopol.2012.12.002
- Carlo, Jessica Luo, James Gaskin, Kalle Lyytinen, and Gregory M. Rose. (2014). "Early vs. Late Adoption of Radical Information Technology Innovations across Software Development Organizations: An Extension of the Disruptive Information Technology Innovation Model." *Information Systems Journal*, Vol. 24 No. 6, pp. 537–569. Doi: 10.1111/isj.12039
- Centobelli, Piera, Roberto Cerchione, and Rajwinder Singh. (2019). "The Impact of Leanness and Innovativeness on Environmental and Financial Performance: Insights from Indian SMEs." *International Journal of Production Economics*, Vol. 212, pp. 111–124. Doi: 10.1016/j.ijpe.2019.02.011
- Chege, Samwel Macharia, Daoping Wang, and Shaldon Leparan Suntu. (2020). "Impact of Information Technology Innovation on Firm Performance in Kenya." *Information Technology for Development*, Vol. 26 No. 2, pp. 316–345. Doi: 10.1080/02681102.2019.1573717
- Chen, Yu-Shan, Tai-Wei Chang, Chun-Yu Lin, Pi-Yu Lai, and Kuan-Hung Wang. (2016). "The Influence of Proactive Green Innovation and Reactive Green Innovation on Green Product Development Performance: The Mediation Role of Green Creativity." *Sustainability*, Vol. 8 No. 10, p. 966. Doi: 10.3390/su8100966
- Chesbrough, Henry William. (2006). "The Era of Open Innovation." *Managing Innovation and Change*, Vol. 127 No. 3, pp. 34–41. Available at: https://books.google.com/books?hl=fr&lr=&id=SDNJzAjyocQC&oi=fnd&pg=PA127&dq=The+Era+of+O pen+Innovation&ots=VfWPHMziJM&sig=6hugOWI6Tg2u9wDhN0YFUWyqSfY
- Chesbrough, Henry William. (2003). Open Innovation: The New Imperative for Creating and Profiting from Technology. Harvard Business Press. Available at: https://books.google.com/books?hl=fr&lr=&id=4hTRWStFhVgC&oi=fnd&pg=PR9&dq=Open+Innovation :+The+New+Imperative+for+Creating+and+Profiting+from+Technology&ots=XuWzTTvaCI&sig=X9yk_KB4-if2-kRfiNQgIHugtEE
- Chung, Hyuk. (2017). "R&D Investment, Cash Holdings and the Financial Crisis: Evidence from Korean Corporate Data." Applied Economics, Vol. 49 No. 55, pp. 5638–5650. Doi: 10.1080/00036846.2017.1319564
- Colombo, Massimo G., Evila Piva, Anita Quas, and Cristina Rossi-Lamastra. (2016). "How High-Tech Entrepreneurial Ventures Cope with the Global Crisis: Changes in Product Innovation and Internationalization Strategies." *Industry and Innovation*, Vol. 23 No. 7, pp. 647–671. Doi: 10.1080/13662716.2016.1196438
- Cuevas-Vargas, Héctor, and Neftalí Parga-Montoya. (2022). "How ICT Usage Affect Frugal Innovation in Mexican Small Firms. The Mediating Role of Entrepreneurial Orientation." *Procedia Computer Science*, Vol. 199, pp. 223–230. Doi: 10.1016/j.procs.2022.01.028
- Czarnitzki, Dirk, and Hanna Hottenrott. (2011). "R&D Investment and Financing Constraints of Small and Medium-Sized Firms." *Small Business Economics*, Vol. 36 No. 1, pp. 65–83. Doi: 10.1007/s11187-009-9189-3
- Damanpour, Fariborz. (2010). "An Integration of Research Findings of Effects of Firm Size and Market Competition on Product and Process Innovations." *British Journal of Management*, Vol. 21 No. 4, pp. 996– 1010. Doi: 10.1111/j.1467-8551.2009.00628.x
- Dass, Parshotam. (2000). "Relationship of Firm Size, Initial Diversification, and Internationalization with Strategic Change." *Journal of Business Research*, Vol. 48 No. 2, pp. 135–146. Doi: 10.1016/S0148-2963(98)00097-6
- De Guimarães, Julio Cesar Ferro, Eliana Andrea Severo, Eric Charles Henri Dorion, Francois Coallier, and Pelayo Munhoz Olea. (2016). "The Use of Organisational Resources for Product Innovation and Organisational Performance: A Survey of the Brazilian Furniture Industry." *International Journal of Production Economics*, Vol. 180, pp. 135–147. Doi: 10.1016/j.ijpe.2016.07.018
- Devaraj, Sarv, and Rajiv Kohli. (2000). "Information Technology Payoff in the Health-Care Industry: A Longitudinal Study." *Journal of Management Information Systems*, Vol. 16 No. 4, pp. 41–67. Doi: 10.1080/07421222.2000.11518265
- Dougherty, Deborah, and Cynthia Hardy. (1996). "Sustained Product Innovation in Large, Mature Organizations:

Overcoming Innovation-to-Organization Problems." Academy of Management Journal, Vol. 39 No. 5, pp. 1120–1153. Doi: 10.5465/256994

- Ebersberger, Bernd, and Andreas Kuckertz. (2021). "Hop to It! The Impact of Organization Type on Innovation Response Time to the COVID-19 Crisis." *Journal of Business Research*, Vol. 124, pp. 126–135. Doi: 10.1016/j.jbusres.2020.11.051
- Enkel, Ellen, and Oliver Gassmann. (2010). "Creative Imitation: Exploring the Case of Cross-Industry Innovation." *R&d Management*, Vol. 40 No. 3, pp. 256–270. Doi: 10.1111/j.1467-9310.2010.00591.x
- Fernández-Portillo, Antonio, Manuel Almodóvar-González, and Ricardo Hernández-Mogollón. (2020). "Impact of ICT Development on Economic Growth. A Study of OECD European Union Countries." *Technology in Society*, Vol. 63, p. 101420. Doi: 10.1016/j.techsoc.2020.101420
- Ferreira, Jorge, Arnaldo Coelho, and Luiz Moutinho. (2020). "Dynamic Capabilities, Creativity and Innovation Capability and Their Impact on Competitive Advantage and Firm Performance: The Moderating Role of Entrepreneurial Orientation." *Technovation*, Vol. 92, p. 102061. Doi: 10.1016/j.technovation.2018.11.004
- Filippetti, Andrea, and Daniele Archibugi. (2011). "Innovation in Times of Crisis: National Systems of Innovation, Structure, and Demand." *Research Policy*, Vol. 40 No. 2, pp. 179–192. Doi: 10.1016/j.respol.2010.09.001
- Frotagheh, Mahdi, and Behzad Kardan. (2019). "The Relationship between Governance Indicators and Firm Performance." Iranian Journal of Accounting, Auditing and Finance, Vol. 3 No. 3, pp. 39–56. Doi: 10.22067/IJAAF.V3I3.89075
- Gago, David, and Luis Rubalcaba. (2007). "Innovation and ICT in Service Firms: Towards a Multidimensional Approach for Impact Assessment." *Journal of Evolutionary Economics,* Vol. 17 No. 1, pp. 25–44. Doi: 10.1007/s00191-006-0030-8
- Galanakis, Charis M., Myrto Rizou, Turki MS Aldawoud, Ilknur Ucak, and Neil J. Rowan. (2021). "Innovations and Technology Disruptions in the Food Sector within the COVID-19 Pandemic and Post-Lockdown Era." *Trends in Food Science & Technology*, Vol. 110, pp. 193–200. Doi: 10.1016/j.tifs.2021.02.002
- Gërguri-Rashiti, Shqipe, Veland Ramadani, Hyrije Abazi-Alili, Léo-Paul Dana, and Vanessa Ratten. (2017). "ICT, Innovation and Firm Performance: The Transition Economies Context." *Thunderbird International Business Review*, Vol. 59 No. 1, pp. 93–102. Doi: 10.1002/tie.21772
- Giebel, Marek, and Kornelius Kraft. (2019). "The Impact of the Financial Crisis on Investments in Innovative Firms." *Industrial and Corporate Change*, Vol. 28 No. 5, pp. 1079-1099. Doi: 10.1093/icc/dty050
- Gomez, Liliana. (2016). "How Does Rule of Law Affect Firm Performance?-Comparing the Effect of Rule of Law in Companies Located in Countries with Different Level of Development." *Journal of Development Economics*, Vol. 85 No. 2, pp. 1–26. Available at: http://arno.uvt.nl/show.cgi?fid=142033
- Graebner, Melissa E., Kathleen M. Eisenhardt, and Philip T. Roundy. (2010). "Success and Failure in Technology Acquisitions: Lessons for Buyers and Sellers." *Academy of Management Perspectives*, Vol. 24 No. 3, pp. 73–92. Doi: 10.5465/amp.24.3.73
- Guillemette, Manon G., and Guy Paré. (2012). "Toward a New Theory of the Contribution of the IT Function in Organizations." *Mis Quarterly*, pp. 529–551. Doi: 10.2307/41703466
- Haggard, Stephan, and Lydia Tiede. (2011). "The Rule of Law and Economic Growth: Where Are We?" *World Development*, Vol. 39 No. 5, pp. 673–685. Doi: 10.1016/j.worlddev.2010.10.007
- Hall, Bronwyn H., Francesca Lotti, and Jacques Mairesse. (2009). "Innovation and Productivity in SMEs: Empirical Evidence for Italy." *Small Business Economics*, Vol. 33 No. 1, pp. 13–33. Doi: 10.1007/s11187-009-9184-8
- Hall, Bronwyn H., Francesca Lotti, and Jacques Mairesse. (2013). "Evidence on the Impact of R&D and ICT Investments on Innovation and Productivity in Italian Firms." *Economics of Innovation and New Technology*, Vol. 22 No. 3, pp. 300–328. Doi: 10.1080/10438599.2012.708134
- Han, Hongfang, and Yanhong Qian. (2021). "Did Enterprises' Innovation Ability Increase During the COVID-19 Pandemic? Evidence From Chinese Listed Companies." Asian Economics Letters, Vol. 1 No. 3, pp. 1–5. Doi: 10.46557/001c.18072
- Hottenrott, Hanna, and Bettina Peters. (2012). "Innovative Capability and Financing Constraints for Innovation: More Money, More Innovation?" *Review of Economics and Statistics*, Vol. 94 No. 4, pp. 1126–1142. Doi: 10.1162/REST_a_00227
- Hu, Shiwei, and Yuyao Zhang. (2021). "COVID-19 Pandemic and Firm Performance: Cross-Country Evidence." International Review of Economics & Finance, Vol. 74, pp. 365–372. Doi: 10.1016/j.iref.2021.03.016
- Huang, Jiashun, Weiping Li, Lijia Guo, and Jim W. Hall. (2021). "Information and Communications Technology Infrastructure and Firm Growth: An Empirical Study of China's Cities." *Telecommunications Policy*, Vol. 46 No. 3, p. 102263. Doi: 10.1016/j.telpol.2021.102263
- Huang, Shi-Ming, Chin-Shyh Ou, Chyi-Miaw Chen, and Binshan Lin. (2006). "An Empirical Study of Relationship between IT Investment and Firm Performance: A Resource-Based Perspective." European

Journal of Operational Research, Vol. 173 No. 3, pp. 984–999. Doi: 10.1016/j.ejor.2005.06.013

- Jackson, Sherry Avery, Venugopal Gopalakrishna-Remani, Rajat Mishra, and Randy Napier. (2016). "Examining the Impact of Design for Environment and the Mediating Effect of Quality Management Innovation on Firm Performance." *International Journal of Production Economics*, Vol. 173, pp. 142–152. Doi: 10.1016/j.ijpe.2015.12.009
- Jimenez, D., and R. Sanz-Valle. (2011). "Innovation, Organizational Learning and Performance." Journal of Business Research, Vol. 64 No. 4, pp. 408-417. Doi: 10.1016/j.jbusres.2010.09.010
- Kaplan, Robert S., and David P. Norton. (1992). "The Balanced Scorecard—Measures That Drive Performance." *Harvard Business Review*, Vol. 83, No 7, p. 172. Available at: https://hbr.org/1992/01/the-balancedscorecard-measures-that-drive-performance-2
- Karlsson, Charlie, and Sam Tavassoli. (2016). "Innovation Strategies of Firms: What Strategies and Why?" *The Journal of Technology Transfer*, Vol. 41 No. 6, pp. 1483–1506. Doi: 10.1007/s10961-015-9453-4
- Klewitz, Johanna, and Erik G. Hansen. (2014). "Sustainability-Oriented Innovation of SMEs: A Systematic Review." *Journal of Cleaner Production*, Vol. 65, pp. 57–75. Doi: 10.1016/j.jclepro.2013.07.017
- Kossaï, Mohamed, and Patrick Piget. (2014). "Adoption of Information and Communication Technology and Firm Profitability: Empirical Evidence from Tunisian SMEs." *The Journal of High Technology Management Research*, Vol. 25 No. 1, pp. 9–20. Doi: 10.1016/j.hitech.2013.12.003
- Kotabe, Masaaki, Crystal X. Jiang, and Janet Y. Murray. (2017). "Examining the Complementary Effect of Political Networking Capability with Absorptive Capacity on the Innovative Performance of Emerging-Market Firms." Journal of Management, Vol. 43 No. 4, pp. 1131–1156. Doi: 10.1177/0149206314548226
- Lawson, Benn, and Danny Samson. (2001). "Developing Innovation Capability in Organisations: A Dynamic Capabilities Approach." *International Journal of Innovation Management*, Vol. 5 No. 03, pp. 377–400. Doi: 10.1142/S1363919601000427
- Lechman, Ewa, and Adam Marszk. (2019). "ICT-Driven Economic and Financial Development: Analyses of European Countries". *Academic Press*, pp. 1-320. Doi: 10.1016/c2017-0-00344-0
- Lee, Haewon, Hanbyeol Choi, Junyeong Lee, Jinyoung Min, and Heeseok Lee. (2016). "Impact of IT Investment on Firm Performance Based on Technology IT Architecture." *Procedia Computer Science*, Vol. 91, pp. 652–661. Doi: 10.1016/j.procs.2016.07.164
- Lee, Neil, Hiba Sameen, and Marc Cowling. (2015). "Access to Finance for Innovative SMEs since the Financial Crisis." *Research Policy*, Vol. 44 No. 2, pp. 370–380. Doi: 10.1016/j.respol.2014.09.008
- Lee, Ryeowon, Jong-Ho Lee, and Tony C. Garrett. (2019). "Synergy Effects of Innovation on Firm Performance." *Journal of Business Research*, Vol. 99, pp. 507–515. Doi: 10.1016/j.jbusres.2017.08.032
- Lee, Sang M., and Silvana Trimi. (2021). "Convergence Innovation in the Digital Age and in the COVID-19 Pandemic Crisis." *Journal of Business Research*, Vol. 123, pp. 14–22. Doi: 10.1016/j.jbusres.2020.09.041
- Leiblein, Michael J., and Tammy L. Madsen. (2009). "Unbundling Competitive Heterogeneity: Incentive Structures and Capability Influences on Technological Innovation." *Strategic Management Journal*, Vol. 30 No. 7, pp. 711–735. Doi: 10.1002/smj.746
- Lichtenthaler, Ulrich. (2016). "Toward an Innovation-Based Perspective on Company Performance." *Management Decision*, Vol. 54 No. 1, pp. 66-87. Doi: 10.1108/MD-05-2015-0161
- Liu, Yang, Jiuyu Dong, Liang Mei, and Rui Shen. (2022). "Digital Innovation and Performance of Manufacturing Firms: An Affordance Perspective." *Technovation*, p. 102458. Doi: 10.1016/j.technovation.2022.102458
- Liu, Yu, Zuobao Wei, and Feixue Xie. (2014). "Do Women Directors Improve Firm Performance in China?" *Journal of Corporate Finance*, Vol. 28, pp. 169–184. Doi: 10.1016/j.jcorpfin.2013.11.016
- Marszk, Adam, and Ewa Lechman. (2021). "Reshaping Financial Systems: The Role of ICT in the Diffusion of Financial Innovations–Recent Evidence from European Countries." *Technological Forecasting and Social Change*, Vol. 167, p. 120683. Doi: 10.1016/j.techfore.2021.120683
- Medase, Stephen Kehinde, and Shoaib Abdul-Basit. (2020). "External Knowledge Modes and Firm-Level Innovation Performance: Empirical Evidence from Sub-Saharan Africa." *Journal of Innovation & Knowledge*, Vol. 5 No. 2, pp. 81–95. Doi: 10.1016/j.jik.2019.08.001
- Meznar, Martin B., and Douglas Nigh. (1995). "Buffer or Bridge? Environmental and Organizational Determinants of Public Affairs Activities in American Firms." Academy of Management Journal, Vol. 38 No. 4, pp. 975–996. Doi: 10.5465/256617
- Nambisan, Satish. (2013). "Information Technology and Product/Service Innovation: A Brief Assessment and Some Suggestions for Future Research." *Journal of the Association for Information Systems*, Vol. 14 No. 4, p. 1. Doi: 10.17705/1jais.00327
- Nandi Santosh, Joseph Sarkis, Aref Aghaei Hervani, and Marilyn M. Helms. (2021). "Redesigning Supply Chains Using Blockchain-Enabled Circular Economy and COVID-19 Experiences." Sustainable Production and Consumption, Vol. 27, pp. 10–22. Doi: 10.1016/j.spc.2020.10.019

- Neirotti and Pesce (2018). "ICT-based innovation and its competitive outcome: the role of information intensity", *European Journal of Innovation Management*, Vol. 22 No. 2, pp. 383-404. Doi: 10.1108/EJIM-02-2018-0039
- Ovuakporie, Oghogho Destina, Kishore Gopalakrishna Pillai, Chengang Wang, and Yingqi Wei. (2021). "Differential Moderating Effects of Strategic and Operational Reconfiguration on the Relationship between Open Innovation Practices and Innovation Performance." *Research Policy*, Vol. 50 No. 1, p. 104146. Doi: 10.1016/j.respol.2020.104146
- Patacconi, Andrea, and Sharon Belenzon. (2014). "Performance, Firm Size, and the'Basicness' of Research." Belenzon S, Patacconi A. "How does firm size moderate firms' ability to benefit from invention? Evidence from patents and scientific publications". European Management Review, Vol. 11 No. 1, pp. 21-45. Available at SSRN: https://ssrn.com/abstract=1446343 or http://dx.doi.org/10.2139/ssrn.1446343
- Paunov, Caroline. (2012). "The Global Crisis and Firms' Investments in Innovation." *Research Policy*, Vol. 41 No. 1, pp. 24–35. Doi: 10.1016/j.respol.2011.07.007
- Piening, Erk P., and Torsten Oliver Salge. (2015). "Understanding the Antecedents, Contingencies, and Performance Implications of Process Innovation: A Dynamic Capabilities Perspective." *Journal of Product Innovation Management*, Vol. 32 No. 1, pp. 80–97. Doi: 10.1111/jpim.12225
- Prajogo, Daniel I. (2016). "The Strategic Fit between Innovation Strategies and Business Environment in Delivering Business Performance." *International Journal of Production Economics*, Vol. 171, pp. 241–249. Doi: 10.1016/j.ijpe.2015.07.037
- Ramadani, Veland, Robert D. Hisrich, Hyrije Abazi-Alili, Léo-Paul Dana, Laxman Panthi, and Lejla Abazi-Bexheti. (2019). "Product Innovation and Firm Performance in Transition Economies: A Multi-Stage Estimation Approach." *Technological Forecasting and Social Change*, Vol. 140, pp. 271–280. Doi: 10.1016/j.techfore.2018.12.010
- Ren-Jye, D., H. H. Hsien, and W. Keisuke. (2018). "Applications of ICTs and Action Recognition for Construction Workers." *Trends Civ. Eng. Archit,* Vol. 1 No. 3, pp. 48–54. Doi: 10.32474/TCEIA.2018.01.000113
- Rivard, Hugues. (2000). "A Survey on the Impact of Information Technology in the Canadian Architecture, Engineering and Construction Industry." J. Inf. Technol. Constr. Vol. 5, pp. 37–56. Doi: 10.1.1.15.308
- Roberts, Nicholas, Galluch, Pamela S., Dinger, Michael, and Grover, Varun. (2012). "Absorptive capacity and information systems research: review, synthesis, and directions for future research", MIS Quarterly, Vol. 36 No. 2, pp. 625-648. Doi: 10.2307/41703470
- Romer, Paul M. (1990). "Endogenous Technological Change." *Journal of Political Economy*, Vol. 98 No. 5, Part 2, pp. 71–102. Doi: abs/10.1086/261725
- Roper, Stephen, and Joanne Turner. (2020). "R&D and Innovation after COVID-19: What Can We Expect? A Review of Prior Research and Data Trends after the Great Financial Crisis." *International Small Business Journal*, Vol. 38 No. 6, pp. 504–514. Doi: 10.1177/0266242620947946
- Rosenbusch, Nina, Jan Brinckmann, and Andreas Bausch. (2011). "Is Innovation Always Beneficial? A Meta-Analysis of the Relationship between Innovation and Performance in SMEs." *Journal of Business Venturing*, Vol. 26 No. 4, pp. 441–457. Doi: 10.1016/j.jbusvent.2009.12.002
- Rousseau, Mary Beth, Blake D. Mathias, Laura T. Madden, and T. Russell Crook. (2016). "Innovation, Firm Performance, and Appropriation: A Meta-Analysis." *International Journal of Innovation Management*, Vol. 20 No. 03, p. 1650033. Doi: 10.1142/S136391961650033X
- Ruzzier, Mitja, and Maja Konecnik Ruzzier. (2015). "On the Relationship between Firm Size, Resources, Age at Entry and Internationalization: The Case of Slovenian SMEs." *Journal of Business Economics and Management*, Vol. 16 No. 1, pp. 52–73. Doi: 10.3846/16111699.2012.745812
- Ryu, Hyun-Sun, and Jae-Nam Lee. (2018). "Understanding the Role of Technology in Service Innovation: Comparison of Three Theoretical Perspectives." *Information & Management*, Vol. 55 No. 3, pp. 294–307. Doi: 10.1016/j.im.2017.08.003
- Sarasvathy, Saras D. (2001). "Causation and Effectuation: Toward a Theoretical Shift from Economic Inevitability to Entrepreneurial Contingency." *Academy of Management Review*, Vol. 26 No. 2, pp. 243– 263. Doi: 10.5465/amr.2001.4378020
- Schryen, Guido. (2013). "Revisiting IS Business Value Research: What We Already Know, What We Still Need to Know, and How We Can Get There." *European Journal of Information Systems*, Vol. 22 No. 2, pp. 139– 169. Doi: 10.1057/ejis.2012.45
- Schumpeter, Joseph. (1934). "The Theory of Economic Development Harvard University Press." *Cambridge, MA*. Available at: https://www.hup.harvard.edu/catalog.php?isbn=9780674879904&content=toc
- Shen, G., and Z. Yuan. (2020). "The Effect of Enterprise Internetization on the Innovation and Export of Chinese Enterprises." *Economic Research Journal*, Vol. 55 No. 1, pp. 33–48.
- Shin, Hakseung, and Juhyun Kang. (2020). "Reducing Perceived Health Risk to Attract Hotel Customers in the

COVID-19 Pandemic Era: Focused on Technology Innovation for Social Distancing and Cleanliness." *International Journal of Hospitality Management*, Vol. 91, p. 102664. Doi: 10.1016/j.ijhm.2020.102664

- Siriram, Rajenlall. (2022). "An Innovative Approach to Overcoming Technological Challenges and Improving Firm Performance." *Technology in Society*, p. 101865. Doi: 10.1016/j.techsoc.2022.101865
- Ssewanyana, Joseph, and Michael Busler. (2007). "Adoption and Usage of ICT in Developing Countries: Case of Ugandan Firms." *International Journal of Education and Development Using ICT*, Vol. 3 No. 3, pp. 49– 59. Available at: http://ijedict.dec.uwi.edu/include/getdoc.php?id=2232&article=349&mode=pdf
- Symeonidis, George. (1996). "Innovation, Firm Size and Market Structure: Schumpeterian Hypotheses and Some New Themes." Doi: 10.1787/18151973
- Tan, Barney, Evelyn Ng, and Junhui Jiang. (2018). "The Process of Technology Leapfrogging: Case Analysis of the National ICT Infrastructure Development Journey of Azerbaijan." *International Journal of Information Management*, Vol. 38 No. 1, pp. 311–316. Doi: 10.1016/j.ijinfomgt.2017.10.008
- Tan, Tih Koon. (2012). "Financial Distress and Firm Performance: Evidence from the Asian Financial Crisis." *Journal of Finance and Accountancy*, Vol. 11, p. 1. Available at: https://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.475.81&rep=rep1&type=pdf
- Teece, David J. (1986). "Profiting from Technological Innovation: Implications for Integration, Collaboration, Licensing and Public Policy." *Research Policy*, Vol. 15 No. 6, pp. 285–305. Doi: 10.1016/0048-7333(86)90027-2
- Teece, David J. (2010). "Technological Innovation and the Theory of the Firm: The Role of Enterprise-Level Knowledge, Complementarities, and (Dynamic) Capabilities." In *Handbook of the Economics of Innovation*, Vol. 1, pp. 679–730. Doi: 10.1016/S0169-7218(10)01016-6
- Teece, David J. (2018). "Profiting from Innovation in the Digital Economy: Enabling Technologies, Standards, and Licensing Models in the Wireless World." *Research Policy*, Vol. 47 No. 8, pp. 1367–1387. Doi: 10.1016/j.respol.2017.01.015
- Torero, Maximo, and Joachim Von Braun. (2006). Information and Communication Technologies for Development and Poverty Reduction: The Potential of Telecommunications. *Intl Food Policy Res Inst.*
- Tran, Quangdung, Changzheng Zhang, Hongyi Sun, and Dechun Huang. (2014). "Initial Adoption versus Institutionalization of E-Procurement in Construction Firms: An Empirical Investigation in Vietnam." *Journal of Global Information Technology Management*, Vol. 17 No. 2, pp. 91–116. Doi: 10.1080/1097198X.2014.928565
- Ueki, Yasushi, and Masatsugu Tsuji. (2019). "The Roles of ICTs in Product Innovation in Southeast Asia." *The Review of Socionetwork Strategies*, Vol. 13 No. 1, pp. 79–95. Doi: 10.1145/3227696.3227719
- Usai, A., F. Fiano, A. Messeni Petruzzelli, P. Paoloni, M. Farina Briamonte, and B. Orlando. (2021). "Unveiling the Impact of the Adoption of Digital Technologies on Firms' Innovation Performance." *Journal of Business Research*, Vol. 133, pp. 327–336. Doi: 10.1016/j.jbusres.2021.04.035
- Vranakis, Stergios K., and Prodromos D. Chatzoglou. (2012). "A Conceptual Model for Machinery & Equipment Investment Decisions." *International Journal of Business and Management*, Vol. 7 No. 1, p. 36. Doi: 10.5539/ijbm.v7n1p36
- Vu, Khuong, Payam Hanafizadeh, and Erik Bohlin. (2020). "ICT as a Driver of Economic Growth: A Survey of the Literature and Directions for Future Research." *Telecommunications Policy*, Vol. 44 No. 2, p. 101922. Doi: 10.1016/j.telpol.2020.101922
- Wang, Yonggui, Aoran Hong, Xia Li, and Jia Gao. (2020). "Marketing Innovations during a Global Crisis: A Study of China Firms' Response to COVID-19." *Journal of Business Research*, Vol. 116, pp. 214–220. Doi: 10.1016/j.jbusres.2020.05.029
- Wang, Zhixin, and Zhenyu Qi. (2021). "Analysis of the Influences of ICTs on Enterprise Innovation Performance in China." *Managerial and Decision Economics*, Vol. 42 No. 2, pp. 474–478. Doi: 10.1002/mde.3247
- Wenzel, Matthias, Sarah Stanske, and Marvin B. Lieberman. (2020). "Strategic Responses to Crisis." *Strategic Management Journal*, Vol. 41 No. 7/18, pp.1-12. Doi: 10.1002/smj.3161
- Xie, Xuemei, Thu Thao Hoang, and Qiwei Zhu. (2022). "Green Process Innovation and Financial Performance: The Role of Green Social Capital and Customers' Tacit Green Needs." *Journal of Innovation & Knowledge*, Vol. 7 No. 1, p. 100165. Doi: 10.1016/j.jik.2022.100165
- Xie, Xuemei, Jiage Huo, and Hailiang Zou. (2019). "Green Process Innovation, Green Product Innovation, and Corporate Financial Performance: A Content Analysis Method." *Journal of Business Research*, Vol. 101, pp. 697–706. Doi: 10.1016/j.jbusres.2019.01.010
- Yeo, Benjamin, and Delvin Grant. (2019). "Exploring the Factors Affecting Global Manufacturing Performance." *Information Technology for Development*, Vol. 25 No. 1, pp. 92–121. Doi: 10.1080/02681102.2017.1315354
- Zhang, Jing A., Conor O'Kane, and Guoquan Chen. (2020). "Business Ties, Political Ties, and Innovation

Performance in Chinese Industrial Firms: The Role of Entrepreneurial Orientation and Environmental Dynamism." *Journal of Business Research*, Vol. 121, pp. 254–267. Doi: 10.1016/j.jbusres.2020.08.055

- Zhang, Yao, Xiaoming Li, Shan Qiao, Yuejiao Zhou, and Zhiyong Shen. (2018). "Information Communication Technology (ICT) Use among PLHIV in China: A Promising but Underutilized Venue for HIV Prevention and Care." International Journal of Information Management, Vol. 38 No. 1, pp. 27–33. Doi: 10.1016/j.ijinfomgt.2017.09.003
- Zhong, Jing, Yonglin Chen, Jiaqi Yan, and Jinlian Luo. (2022). "The Mixed Blessing of Cyberloafing on Innovation Performance during the COVID-19 Pandemic." *Computers in Human Behavior*, Vol. 126, p. 106982. Doi: 10.1016/j.chb.2021.106982
- Zhou, Qingna, Ping Gao, and Admos Chimhowu. (2019). "ICTs in the Transformation of Rural Enterprises in China: A Multi-Layer Perspective." *Technological Forecasting and Social Change*, Vol. 145, pp. 12–23. Doi: 10.1016/j.techfore.2019.04.026
- Zhu, Facang, Qianqian Li, Shichun Yang, and Tomas Balezentis. (2021). "How ICT and R&D Affect Productivity? Firm Level Evidence for China." *Economic Research-Ekonomska Istraživanja*, pp. 1–19. Doi: 10.1080/1331677X.2021.1875861
- Zhu, Xiaoxuan, Zhenxin Xiao, Maggie Chuoyan Dong, and Jibao Gu. (2019). "The Fit between Firms' Open Innovation and Business Model for New Product Development Speed: A Contingent Perspective." *Technovation*, Vol. 86, pp. 75–85. Doi: 10.1016/j.technovation.2019.05.005